UBC

SUBJECT	Proposed Policy Amendment COI Policy (SC3)		
SUBMITTED TO	Employee Relations Committee		
MEETING DATE	March 13, 2024		
SESSION CLASSIFICATION	Recommended session criteria from Board Meetings Policy: OPEN		
REQUEST	Action requested - Recommendation to Board for approval		
	IT IS HEREBY RESOLVED that the Employee Relations Committee recommends that the Board of Governors:		
	 Approve amendments to the COI Policy (SC3) ("Proposed Amended COI Policy (SC3)") in the form set out in Appendix 1 attached to this briefing, effective as of April 2, 2024 ("Effective Date"), provided however that any investigation or disciplinary proceedings for any alleged or actual breaches of the current COI Policy commenced prior to the Effective Date will continue to be governed under the current COI Policy. 		
LEAD EXECUTIVE	Hubert Lai, K.C., University Counsel (Responsible Executive)		
SUPPORTED BY	Chelsea Thompson, Legal Counsel (Policy Development Committee Chair)		
	Jeffrey Toward, COI Manager (Policy Development Committee Secretary)		

PRIOR SUBMISSIONS

The subject matter of this submission has been considered previously by the Employee Relations Committee on the following occasion:

1. November 20, 2023 (OPEN SESSION)

Action/Follow up: The Employee Relations Committee considered the proposed amendments to the COI Policy, following which the Office of the University Counsel ("**OUC**") published the proposed amendments for community consultation.

The following Executive Summary assumes familiarity with the prior submission and provides a status update from the date of the prior submission.

EXECUTIVE SUMMARY

Following the Employee Relations Committee's consideration of the proposed amendments to the COI Policy during its meeting on November 20, 2023, the OUC published a notice of the proposed amendments and invited community feedback on the amended COI Policy proposal. The comment period remained open from November 21, 2023 until January 21, 2024.

Calls for comments were also published in the November 28, 2023 and January 9, 2024 editions of UBC Today, the weekly newsletter that is published on UBC's publicly-facing web site and which is also distributed by e-mail to all UBC faculty members and staff. Email notices were also sent out on November 21, 2023 and January 15, 2024 to everyone on the open self-service subscription service maintained by the OUC, which includes representatives from the Alma Mater Society, the Graduates Students Society, and all employee associations and unions (other than BCGEU child care and BCGEU Okanagan, who have elected not to receive subscription e-mails).

The OUC received 18 specific comments from four sources, including both faculty and staff members. The Policy Development Committee reconvened to consider the feedback and to update its recommended amendments to the COI Policy as appropriate.

A table setting out the community input that was received and the Policy Development Committee's response to each comment can be found in Supplemental Materials #4. The full text of the Proposed Amended COI Policy (SC3), highlighted to show changes from the version that was published for community consideration and feedback, is attached as Supplemental Materials #1.

The key changes that have been made to the Proposed Amended COI Policy (SC3) and associated Procedures in response to the community feedback are as follows:

- Amend section 2.2.1 of the Policy to clarify what activities in addition to those included under the definition of Non-University Activities may trigger a Conflict of Commitment or Conflict of Interest, so that the Policy now refers, where appropriate, to Non-University Activities rather than the generic use of the word "activity" or "activities" as the definition of "Non-University Activities" is comprehensive and includes any activity outside a UBC Person's scope of work with UBC.
- Amend section 2.2.1 of the Policy and Sections 3.1.1 and 3.2.4 of the Procedures to clarify that not only does a UBC Person have an obligation to immediately disclose a Conflict of Interest or Conflict of Commitment if they intend to engage in any Non-University Activities but a continuing obligation to disclose if there is knowledge of any conflicts that arise during the course of the UBC Person engaging in any Non-University Activities.
- 3. Amend section 2.2.3 of the Policy to clarify that a Faculty Member must disclose and then obtain written approval from the Dean of the faculty, not just their Initial Reviewer, if they intend to engage in an Outside Professional Activity for more than that 52 days per year or if their involvement in an Outside Professional Activity will keep them away from their place of work at UBC for a period of 30 consecutive days, not only in circumstances where they exceed the number of prescribed days that the Faculty Member is permitted to be involved in Outside Professional Activities.
- 4. Amend section 4.3 of the Policy to include language to permit the collection and release of anonymized statistics or summaries to the community for the purpose of education, training or promoting compliance with the Policy.
- 5. Amend section 1.2 of the Procedures to modify the definition of "Business" to remove the exemption for affiliated health authorities and UBC controlled entities, in order to expand the list of activities that need to be declared, ensuring that Initial Reviewers have a more comprehensive declaration and will be able to make more informed decisions about any Conflicts of Interest or Conflicts of Commitment.
- 6. Amend section 3.3.1 of the Procedures to provide additional clarity and ensure consistency across the disclosures made by both Staff and Faculty Members, language has been amended in the Procedures that address the responsibility to disclose all Non-University Activities for both groups of individuals.

- 7. Amend section 10.2 of the Procedures to clarify that, at the direction of the Dean or Initial Reviewer, either the Office of Research Services or the University-Industry Liaison Office may freeze the research accounts of faculty members who fail to comply with the Policy. It has also been noted that in such circumstances UBC may terminate access to funding, as opposed to terminating the funding itself.
- 8. Amend section 2.1.7 of the Procedures to ensure that activities related to human resources decisions where a Conflict of Interest is deemed to exist, specifically include any appointment or selection process, as well as discipline and investigations.

The Proposed Amended COI Policy (SC3) is supported and recommended by the Policy Development Committee and the Responsible Executive for the COI Policy, being the University Counsel.

APPENDICES

1. Proposed Amended COI Policy (SC3) being presented for Board of Governors' approval

SUPPLEMENTAL MATERIALS (optional reading for Governors)

- 1. Blackline of proposed amendments to the COI Policy showing changes against the version previously presented to the Employee Relations Committee for information in November 2023
- 2. Blackline of proposed amendments to the COI Policy showing changes against the current version
- 3. Current Version of the COI Policy
- 4. Summary of community feedback and Policy Development Committee recommendations

UBC	The University of British Columbia Board of Governors	Policy No.: SC3	
Long Title:	Conflict of Interest and Conflict of Commitment		
Short Title:	COI Policy		

Background & Purposes:

UBC's mission and mandate as a public institution of higher education is the advancement and dissemination of knowledge. UBC encourages all UBC Persons to seek and participate in research, to consult widely, and to engage in activities, including Non-University Activities, in a manner that does not interfere with their obligations to UBC.

Occasionally, UBC Persons may find themselves in either a Conflict of Interest or a Conflict of Commitment. The existence of a Conflict of Interest or Conflict of Commitment does not necessarily imply wrongdoing by a UBC Person. Such conflict(s) may exist regardless of a UBC Person's character, intentions, or motivations. However, Conflicts of Interest or Conflicts of Commitment that go unnoticed or are improperly managed may harm the reputation and integrity of the UBC Person(s) and UBC; and may also undermine the public's confidence in the abilities of the UBC Person(s) and UBC to pursue and disseminate knowledge, devoid of bias and personal interests. Without that public confidence, the effectiveness of UBC as a public institution and of UBC Persons is diminished.

The purposes of this Policy are to identify:

- situations that may give rise to a Conflict of Interest or Conflict of Commitment;
- a UBC Person's responsibility to identify, disclose, assess, and manage, if appropriate, a Conflict of Interest or Conflict of Commitment;
- the process for investigating any Conflict of Interest or Conflict of Commitment; and
- disciplinary actions that may result from a violation of this Policy or any associated Procedures or Rules.

Capitalized terms used in this Policy that are not otherwise defined herein shall have the meanings given to such terms in any associated Procedures or Rules.

1. General

1.1 All UBC Persons are responsible for informing themselves and complying with:

- 1.1.1 the requirements of this Policy and any associated Procedures and Rules; and
- 1.1.2 other applicable conflicts of interest or conflicts of commitment policies imposed by other third-party organizations, agencies, or institutions related to their activities at UBC.
- 1.2 All Faculty Members are responsible for informing themselves and complying with the requirements of any Tri-Agency or the U.S. Department of Health and Human Services; internal or external governance bodies (e.g., UBC's research ethics boards, UBC's Animal Care Committee, regulatory colleges such as the College of Physicians and Surgeons of British Columbia) that authorize research; other funding and granting agencies and organizations or third-party contracts, if related to their activities at UBC.
- 1.3 All UBC Persons are responsible for ensuring that participation in a UBC related activity and any Non-University Activity by a UBC Person is undertaken in compliance with this Policy, including ensuring such activities are conducted in a manner that:
 - 1.3.1 is consistent with UBC's mission and mandate to advance and disseminate knowledge;
 - 1.3.2 maintains the general public's trust and confidence that UBC and UBC Persons are able to undertake UBC's mission and mandate devoid of bias and personal interest; and
 - 1.3.3 is undertaken by UBC Persons acting with integrity and adhering to the highest ethical standards at all times, including declaring and appropriately managing any Conflict of Interest or Conflict of Commitment in a timely, open, forthright, constructive and accountable manner.

2. Conflicts of Commitment

2.1 Identifying Conflicts of Commitment

- 2.1.1 All UBC Persons are responsible for identifying, disclosing, and appropriately managing any Conflict of Commitment in accordance with this Policy and conducting themselves with the highest ethical standards, including considering any Conflict of Commitment in a manner that will bear the closest scrutiny.
- 2.1.2 UBC has not attempted to develop an exhaustive list of Conflicts of Commitment as each situation is dependent upon its specific facts. However, the following are examples of situations where a Conflict of Commitment is deemed to exist:
 - (a) where a UBC Person engages in a Non-University Activity during their normal work hours at UBC;
 - (b) where a UBC Person employed on a full-time basis by UBC enrolls in adegree or diploma program at UBC, or any other educational institution;
 - (c) where a UBC Person uses UBC resources for Non-University Activities;

- (d) where a full-time Faculty Member's total involvement in Outside Professional Activities is greater than 52 days per year, including evenings, weekends and vacations; or
- (e) where a full-time Faculty Member will be away from their place of work at UBC for a period of 30 consecutive days (excluding holidays) as a result of the Faculty Member's Outside Professional Activities.

2.2 **Disclosing Conflicts of Commitment**

- 2.2.1 If a UBC Person intends to engage in any Non-University Activities that could lead to a Conflict of Commitment, then the UBC Person must immediately disclose that activity to their Initial Reviewer pursuant to Section 3 of the Procedures. These disclosure obligations also apply to all UBC Persons as soon as knowledge of the Conflict of Commitment arises and continue until the Conflict of Commitment ceases to exist.
- 2.2.2 In addition, all Faculty Members and Staff Members are required to file annual Conflict of Commitment disclosures.
- 2.2.3 If a Faculty Member intends to engage in any activity set out in Section 2.1.2(d) or Section 2.1.2(e) of this Policy, then they must disclose and obtain prior written authorization from the Dean of the faculty in which they hold their primary appointment, a copy of which must be filed with the COI Manager.
- 2.2.4 All UBC Persons are encouraged to bring any questions or concerns regarding their Conflicts of Commitment to their Initial Reviewer, who may consult with the COI Manager.

2.3 Assessing and Managing Conflicts of Commitment

- 2.3.1 If an Initial Reviewer becomes aware of a Conflict of Commitment, whether through disclosure or otherwise, then they must assess the Conflict of Commitment taking into consideration the following:
 - (a) whether a UBC Person's engagement in the Non-University Activity interferes with their professional obligations to UBC, including the elements set forth in Section 1.3 of this Policy; and
 - (b) whether the benefit to UBC from a UBC Person's engagement in the Non-University Activity outweighs the interference with their professional obligations to UBC or would outweigh the interference if appropriately managed through a Management Plan.
- 2.3.2 Once the Initial Reviewer has assessed the Conflict of Commitment situation, the Initial Reviewer, may determine what action, if any, is appropriate pursuant to Section 4 of the Procedures.

3. Conflicts of Interest

3.1 Identifying Conflicts of Interest

- 3.1.1 All UBC Persons are responsible for identifying, disclosing, and appropriately managing any Conflict of Interest in accordance with this Policy and conducting themselves at all times with the highest ethical standards, including considering any Conflict of Interest in a manner that will bear the closest scrutiny.
- 3.1.2 Any UBC Person who engages in Non-University Activities is responsible for identifying any Conflict of Interests arising naturally from their engagement in Non-University Activities.
- 3.1.3 Identifying a Conflict of Interest involves two elements:
 - (a) the UBC Person's or a Related Party's personal interest(s), including without limitation, business, commercial or Financial Interests and personal matters and career interests; and
 - (b) the UBC Person's professional obligations to UBC.
- 3.1.4 A Conflict of Interest includes the following situations:
 - (a) an actual Conflict of Interest;
 - (b) a situation where the UBC Person's personal interest has not yet conflicted but has the potential or might be expected to conflict with the UBC Person's professional obligations to UBC; or
 - (c) the perception of a Conflict of Interest from the perspective of a reasonably wellinformed, impartial observer.
- 3.1.5 UBC has not attempted to develop an exhaustive list of Conflicts of Interest as each situation is dependent upon its specific facts. However, a non-exhaustive list of situations where a Conflict of Interest is deemed to exist is set forth in Section 2 of the Procedures.

3.2 Disclosing Conflicts of Interest

3.2.1 If a UBC Person intends to engage in an activity that could lead to a Conflict of Interest, then the UBC Person must immediately disclose that activity to their Initial Reviewer pursuant to Section 3 of the Procedures and to all parties affected by the Conflict of Interest. For example, if a UBC Person makes a public statement, in the media or otherwise, and identifies themselves as being associated with UBC, then the UBC Person must disclose, as part of that public statement, any other relevant relationships, circumstances, or engagements in other Non-University Activities that give rise to a Conflict of Interest.

- 3.2.2 If prior disclosure is impossible, then disclosure must be made as soon as possible after commencement of the activity.
- 3.2.3 Prior to commencing a research project, a UBC Person who is a principal investigator or researcher must disclose all Conflicts of Interest that may arise in relation to the research project.
- 3.2.4 These disclosure obligations also apply to all UBC Persons as soon as knowledge of the Conflict of Interest arises and continue until the Conflict of Interest ceases to exist.
- 3.2.5 In addition, all Faculty Members and Staff Members are required to file annual Conflict of Interest disclosures.
- 3.2.6 All UBC Persons are encouraged to bring any questions or concerns regarding their Conflicts of Interest to their Initial Reviewer, who may consult with the COI Manager.

3.3 Assessing and Managing Conflicts of Interest

- 3.3.1 If an Initial Reviewer becomes aware of a Conflict of Interest, whether through disclosure or otherwise, then they must assess the situation to determine what action, if any, is appropriate, including taking into consideration the following:
 - (a) the impact on the UBC Person's ability to satisfy their professional obligations to UBC, including the elements set forth in Section 1.3 of this Policy;
 - (b) the degree to which the proposed action will be detrimental to the interests of UBC;
 - (c) the degree to which it may harm the reputation and integrity of the UBC Person(s) or UBC, or both;
 - (d) the degree to which it may compromise the professional judgment of any UBC Person involved in conducting or reporting research;
 - (e) the extent to which the proposed action or activity could be managed through an appropriate Management Plan; and
 - (f) any of the other Conflict of Interest elements set forth in Section 2 of the Procedures.
- 3.3.2 Once the Initial Reviewer has assessed the Conflict of Interest situation, the Initial Reviewer, may determine what action, if any, is appropriate pursuant to Section 4 of the Procedures.

4. Reporting of Complaints or Concerns

4.1 All persons are encouraged to contact the COI Manager in writing if they believe that a Conflict of Interest or Conflict of Commitment has been unreported, misrepresented, unassessed, or

unmanaged or that there has been any other breach of this Policy or any associated Procedures or Rules. Upon receipt, the COI Manager will handle the complaint or concern in accordance with Section 6.1.4 of the Procedures.

- 4.2 All complaints or concerns submitted pursuant to this Policy will be taken seriously. The anonymity of the person making the complaint or raising a concern will, unless otherwise directed by that person, be maintained, except as required by a legal process or the law. However, any information provided anonymously cannot be acted upon unless the information is independently substantiated. Allegations of retaliation against any person who, in good faith, makes a complaint or raises a concern pursuant to this Policy, gives evidence, or otherwise participates in a process under this Policy will be handled in accordance with UBC's Retaliation Policy.
- 4.3 UBC will protect the personal information of all parties involved as required under applicable legislation, including the *Freedom of Information and Protection of Privacy Act* (British Columbia). Except as required by a legal process or law, any personal information disclosed by a UBC Person pursuant to this Policy will be held in confidence and will only be available to relevant granting or funding agencies and those with a need to know the information in order to fulfill their responsibilities to UBC, including:
 - 4.3.1 those responsible under this Policy and any associated Procedures or Rules for the review, management, or audit of Conflict of Interest and Conflict of Commitment disclosures; or
 - 4.3.2 those who are involved in investigations of alleged breaches of UBC Policies or other misconduct, which may include: Central Human Resources; the Office of the Vice President, Research and Innovation; the Office of the University Counsel; members of the COI Committee; the Responsible Executive; or any of their respective and applicable delegates.

PROCEDURES ASSOCIATED WITH THE POLICY



Pursuant to the Regulatory Framework Policy, the President may approve Procedures or the amendment or repeal of Procedures. Such approvals must be reported at the next meeting of the UBC Board of Governors or as soon thereafter as practicable.

Capitalized terms used in the Policy that are not otherwise defined herein shall have the meanings given to such terms in these Procedures or Rules.

1. **Definitions**

- 1.1 "Administrative Head(s) of Unit" means a Director of an administrative unit; a Head of an academic department; a Director of a centre, institute or school; a Principal of a college; a Dean of a faculty; the Registrar; the University Librarian; a Vice-President; or the President or the equivalent; or any of their respective delegates, which may include Associate Vice-Presidents; Vice-Principals; and Associate Vice-Principals.
- 1.2 **"Business**" means a corporation, partnership, sole proprietorship, firm, franchise, association, organization, holding company, joint stock company, receivership, business or real estate trust, or other entity organized for profit, not for profit, or charitable purposes.
- 1.3 "Central Human Resources" means Employee and Labour Relations or Faculty Relations on the UBC Vancouver campus, or Human Resources on the UBC Okanagan campus.
- 1.4 "COI Committee" means a committee established by the Responsible Executive to carry out the roles and responsibilities set out in the Policy, and which includes at least one independent person who is not a UBC Person.
- 1.5 **"COI Manager**" means the individual appointed by the Responsible Executive to oversee the administration and governance of the Policy or other duties as delegated by the Responsible Executive from time to time, and, whose contact information can be found on the website of the office of the Responsible Executive.
- 1.6 "Conflict(s) of Commitment" means a situation where a UBC Person engages in Non-University Activities that adversely affect the discharge of the UBC Person's primary responsibilities to UBC.
- 1.7 "**Conflict(s) of Interest**" means a situation in which a UBC Person, or their Related Party(ies), has a personal interest that actually conflicts, could conflict, or there is a perception that it conflicts with the UBC Person's obligations to UBC.
- 1.8 **"Faculty Member**" means an individual appointed by the University in the following faculty titles, ranks and descriptions: professor, associate professor, assistant professor, professor of teaching, assistant professor of teaching, associate professor of teaching, adjunct professor, clinical professor, clinical associate professor, clinical instructor,

clinical fellow, academic program director, librarian, research associate, post-doctoral fellow, sessional lecturer, lecturer, visiting assistant professor, visiting associate professor, visiting professor, visiting lecturer, visiting scientist, visiting scholar, emeritus, honorary faculty or affiliate appointment or in an equivalent position designated by the applicable senate.

- 1.9 "Financial Interest" means having or having the expectation to receive:
 - 1.9.1 greater than a 5% ownership interest in a Business; or
 - 1.9.2 anything with a monetary value exceeding \$5,000 in any one calendar year, including remuneration (e.g., salary, consulting fees, retainers, honoraria, bonuses, gifts, speaker's fees, advisory board remuneration, finders or recruitment fees), equity interests (e.g., stocks, stock options or other ownership interests), and intellectual property rights (e.g., patents, copyrights, royalties or other payments from such rights).
- 1.10 "Guidelines" has the meaning set forth in UBC's Regulatory Framework Policy.
- 1.11 "Initial Reviewer(s)" means the individual responsible for reviewing a UBC Person's disclosures of Conflicts of Interest or Conflicts of Commitment. For Faculty Members, the Initial Reviewer is normally the Dean of the faculty in which they hold their primary appointment, unless their faculty is departmentalized, in which case it is normally the Head of their department or school. For Staff Members, the Initial Reviewer is normally their immediate supervisor and, where the Staff Member has more than one immediate supervisor, the Initial Reviewer is the immediate supervisor designated by the Responsible Executive. For all other UBC Persons, the Initial Reviewer is normally their immediate supervisor. Notwithstanding the foregoing, the Responsible Executive may designate a different Initial Reviewer for any UBC Persons.
- 1.12 "Management Plan" means the written document governing and managing a UBC Person's participation in an activity deemed to constitute a Conflict of Interest or Conflict of Commitment.
- 1.13 "Non-University Activity(ies)" means any activity outside a UBC Person's scope of work with UBC and includes Outside Professional Activities.
- 1.14 "Outside Professional Activity(ies)" means any activity outside a UBC Person's scope of work with UBC that involves the same specialized skill and knowledge that the UBC Person utilizes in their work with UBC and includes the operation of a Business, consulting or advisory services, external teaching, external academic appointments and external speaking engagements.
- 1.15 "Policy" means the COI Policy.
- 1.16 **"Position of Authority"** means any position that includes responsibilities for a material function of the operation or management of a Business.
- 1.17 "Related Party(ies)" means a UBC Person's immediate family member (e.g., spouse, partner, or sibling), direct descendant or ancestor (e.g., child, grandchild, parent, or grandparent), any other person living in the same household, any other person with whom the UBC Person shares a Financial Interest, either directly or indirectly, or any Business in which the UBC Person has an ownership interest of more than 5%.

- 1.18 "**Related Policy(ies)**" means the related policies specific to the Policy as identified in the official repository for the Policy maintained by the Office of the University Counsel and published on its website.
- 1.19 "**Respondent**" means the UBC Person who is the subject of an investigation of an alleged breach of the COI Policy or associated Procedures or Rules.
- 1.20 **"Responsible Executive"** means individual(s) assigned by the President, pursuant to UBC's Regulatory Framework Policy, to be responsible for the Policy; and any sub-delegate of that assigned responsible individual(s) except to the extent that the power to delegate is specifically excluded in the Policy or in the appointment by the President.
- 1.21 "Rules" has the meaning set forth in UBC's Regulatory Framework Policy.
- 1.22 "**Staff Member**" means individual(s) employed by UBC in any of the following employee categories or titles:
 - 1.22.1 Management and Professional;
 - 1.22.2 Director of a service unit;
 - 1.22.3 Administrative AVP (as defined in UBC's Administrative AVPs Policy);
 - 1.22.4 Administrative VP (as defined in UBC's Administrative VPs Policy); and
 - 1.22.5 Any other UBC Person(s), who is not a Faculty Member, as required by the Responsible Executive (e.g., those who have signing authority over UBC financial accounts).
- 1.23 "**Tri-Agency**" means the Canadian Institutes of Health Research, the Natural Sciences and Engineering Research Council, and the Social Sciences and Humanities Research Council, collectively.
- 1.24 "UBC Person(s)" means all Faculty Members, all Staff Members, all other employees of UBC, all students to the extent that they are carrying out UBC research, and all individual(s) who have agreed to carry out duties on behalf of UBC as a volunteer.

2. Additional Conflict of Interest Considerations:

- 2.1 The following is a non-exhaustive list of situations where a Conflict of Interest is deemed to exist:
 - 2.1.1 where a UBC Person's responsibility to instruct, evaluate and supervise students in a fair, unbiased and effective manner is or could be impeded or compromised (Note, the inherent power imbalance that exists between a UBC Person and a student must not be used for personal benefit.);
 - 2.1.2 where a UBC Person or a Related Party has a Financial Interest in the outcome of their research, including any Financial Interest in a company sponsoring research at their UBC laboratory or facilities;

- 2.1.3 where a UBC Person or a Related Party has a Financial Interest in their teaching activities at UBC, other than their annual salary from UBC;
- 2.1.4 where a UBC Person has influence over a decision about a proposed relationship between UBC and a Business in which the UBC Person or a Related Party has a Financial Interest or holds a Position of Authority;
- 2.1.5 where a UBC Person or a Related Party has or obtains a Financial Interest or a Position of Authority in a Business with which UBC has an existing relationship and the Business is related to the UBC Person's work at UBC;
- 2.1.6 where a UBC Person transfers research results or intellectual property arising from their UBC research to a Business in which the UBC Person or a Related Party has a Financial Interest or holds a Position of Authority;
- 2.1.7 where a UBC Person is in a position to influence human resource decisions (such as recruitment, offer of employment, appointment or selection process, evaluation of performance, promotion, granting of tenure, discipline, investigations, or termination of employment) or admission decisions with respect to a person who is a Related Party or who has another relationship with the UBC Person that might reasonably be perceived as creating a Conflict of Interest;
- 2.1.8 where a UBC Person uses or attempts to use their position with UBC to solicit students, other UBC Persons, government agencies, private companies, or members of the public for Non-University Activities, including employing students or other UBC Persons to perform services for the UBC Person or a Related Party on personal or commercial matters;
- 2.1.9 where a UBC Person uses information that is acquired as a result of their position with UBC and not in the public domain for Non-University Activities, unless the UBC Person has proprietary rights (usually enforceable through copyright) to that information (Note, UBC Persons should also be aware that insider trading restrictions may also apply to them under the laws of British Columbia or other jurisdictions.);
- 2.1.10 where a UBC Person's obligations to a board of directors, advisory boards, or the like of an outside organization interfere with or compromise the UBC Person's obligations to UBC;
- 2.1.11 where a UBC Person accepts gifts with a value of \$500 or more that would not have been offered if not for the UBC Person's position or role at UBC;
- 2.1.12 where a Faculty Member enrolls as a student in their own faculty; or
- 2.1.13 where personal considerations could be reasonably perceived to compromise a UBC Person's professional judgment in conducting or reporting research, teaching, or carrying out administrative activities at UBC.

3. Required Disclosures

- 3.1 All UBC Persons must:
 - 3.1.1 immediately disclose to their Initial Reviewer if a UBC Person intends to engage in any Non-University Activities that could lead to a Conflict of Interest and Conflict of Commitment;
 - 3.1.2 immediately disclose Conflicts of Interest and Conflicts of Commitment to their Initial Reviewer as soon as such situation or event arises; and
 - 3.1.3 continue to disclose and submit such information, as requested, or at least annually until the Conflicts of Interest and Conflicts of Commitment cease to exist.
- 3.2 At least annually, all Faculty Members and Staff Members must submit a completed disclosure that discloses the following:
 - 3.2.1 all Non-University Activities;
 - 3.2.2 all circumstances (if any) that give rise to a Conflict of Interest or Conflict of Commitment; and
 - 3.2.3 such additional information asked for in the prescribed annual disclosure form, as it may be amended from time to time.
- 3.3 All disclosures made by Faculty Members under Section 3.2 of these Procedures must be submitted utilizing the Conflict of Interest or the Conflict of Commitment module in the *RISe* system.
- 3.4 All disclosures made by Staff Members under Section 3.2 of these Procedures must be submitted utilizing the Conflict of Interest or the Conflict of Commitment module in the *Workday* system.
- 3.5 All disclosures made by UBC Persons under Section 3.1 of these Procedures who are not Faculty Members or Staff Members must be submitted utilizing the Conflict of Interest or Conflict of Commitment module in the system deemed most appropriate by the Initial Reviewer.

4. Role of Initial Reviewer

- 4.1 The Initial Reviewer is responsible for assessing the disclosure and determining whether there are any Conflicts of Interest or Conflicts of Commitment, and, if so, whether they can and should be permitted subject to a Management Plan.
- 4.2 If the Initial Reviewer determines that a proposed action or activity, including a proposed research project, is:
 - 4.2.1 permissible as disclosed with no further action, then the Initial Reviewer's determination will be recorded in writing and the UBC Person may proceed with the

action or activity subject to obtaining all other applicable approvals;

- 4.2.2 permissible only if appropriately managed through the implementation of a Management Plan, then the Initial Reviewer will develop and implement a Management Plan to manage the Conflict of Interest or the Conflict of Commitment and the UBC Person may proceed with the action or activity pursuant to the Management Plan; or
- 4.2.3 not permissible, then the UBC Person must not proceed with the proposed action or activity giving rise to the Conflict of Interest or Conflict of Commitment.
- 4.3 The Initial Reviewer may consult with the COI Manager in relation to any decisions required to be made under Section 4 of these Procedures, including the development of any applicable Management Plan(s) and the COI Manager may in turn consult with the COI Committee.
- 4.4 If the Initial Reviewer is unable to determine whether a Conflict of Interest or a Conflict of Commitment or any proposed action or activity is permissible due to their own Conflict of Interest or otherwise, then the Initial Reviewer must refer the disclosure one level up for handling. If such a referral is not appropriate, then the matter will be referred to the Responsible Executive for a final determination of who should handle the matter.
- 4.5 Where the UBC Person is a Faculty Member who holds an appointment in more than one faculty, the Initial Reviewer must keep the Dean of the other faculty up-to-date with a copy of that Faculty Member's disclosures and any applicable Management Plan(s). Where the UBC Person is a Staff Member who has more than one immediate supervisor, then the Initial Reviewer must consult with and keep the other immediate supervisors up-to-date with a copy of that Staff Member's disclosures and any applicable Management Plan(s).

5. Unit-Specific Review Process

5.1 Notwithstanding Section 4 of these Procedures, if the UBC Person disclosing a Conflict of Interest or a Conflict of Commitment is part of an academic or administrative unit for which unit-specific review processes have been established pursuant to one or more Rules, then the review shall be undertaken in accordance with such Rules.

6. Role of the COI Manager

- 6.1 The COI Manager is charged with the following responsibilities:
 - 6.1.1 serving as the secretary to the COI Committee;
 - 6.1.2 providing advice and guidance to Initial Reviewers, Administrative Heads of Unit, the Responsible Executive, and the COI Committee with respect to any matter pertaining to the Policy and any associated Procedures and Rules, including without limitation:
 - (a) the establishment of processes for the assessment of Conflicts of Interest and Conflicts of Commitment;

- (b) the assessment of disclosures of Conflicts of Interest and Conflicts of Commitment; and
- (c) the development of appropriate Management Plan(s);
- 6.1.3 making recommendations to the Responsible Executive in relation to the issuance of any Rules, Guidelines, templates, or other materials or resources and consulting the COI Committee as appropriate;
- 6.1.4 handling any complaints or concerns received pursuant to Section 4.1 of the Policy as the COI Manager deems appropriate, including forwarding such complaints or concerns:
 - (a) to the appropriate Initial Reviewer or any other person identified in a Rule establishing a unit-specific review process contemplated by Section 5 of these Procedures for assessment and management pursuant to either Section 2.3 or Section 3.3 the Policy; or
 - (b) to the attention of the COI Committee for handling pursuant to Section 9 of these Procedures; and
- 6.1.5 providing training to UBC Persons regarding Conflicts of Interest, Conflicts of Commitment, and the application of the Policy and any associated Procedures and Rules.

7. Role of the COI Committee

- 7.1 The Responsible Executive must establish a COI Committee.
- 7.2 The COI Committee is charged with the following responsibilities:
 - 7.2.1 auditing random samples of Conflicts of Interest and Conflicts of Commitment that have been approved and any related Management Plans for quality control purposes and advising Initial Reviewers on appropriate standards for the quality of disclosures, reviews, and Management Plans;
 - 7.2.2 acting as a sounding board and source of advice regarding the appropriate handling of specific disclosures or other cases referred to it by the COI Manager.
 - 7.2.3 serving as the final avenue of appeal (such as where a UBC Person disagrees with the assessment of an Initial Reviewer or the assessment resulting from a Rule established under Section 11 of these Procedures);
 - 7.2.4 to review concerns or issues in accordance with Section 9 of these Procedures, whether self-initiated or brought forth by the COI Manager pursuant to Section 6.1.4(b) of these Procedures; and

7.2.5 fulfilling such other roles and responsibilities as may be assigned to it by the Responsible Executive.

8. Appeals

- 8.1 All UBC Persons may appeal any decision regarding a Conflict of Interest or Conflict of Commitment made by their Initial Reviewer in accordance with Section 4 of these Procedures by sending a notice of appeal to the COI Manager in writing using such form as may be prescribed by the COI Manager. The COI Manager will bring the notice to the attention of the COI Committee.
- 8.2 The COI Committee may, at its discretion, refuse to hear an appeal that is filed more than 30 days after the date of the Initial Reviewer's decision.
- 8.3 The Chair of the COI Committee has the discretion to determine the manner in which the COI Committee considers appeals, provided that such discretion must not be exercised in a manner that is inconsistent with the Policy or any associated Procedures or Rules or any directions issued by the Responsible Executive.

9. Referral of Complaints or Concerns to the COI Committee

- 9.1 If a complaint or concern is forwarded to the attention of the COI Committee by the COI Manager pursuant to Section 6.1.4(b) of these Procedures, the COI Committee will normally make an initial determination as to whether the alleged breach has resulted from an honest and reasonable error or oversight. The COI Committee may reconsider its initial determination at any time.
- 9.2 Where the COI Committee's initial determination is that the alleged breach has resulted from an honest and reasonable error or oversight, the COI Committee will normally advise or direct the COI Manager, the Respondent, their Initial Reviewer and any other person that the COI Committee determines should receive such advice or directions as to the appropriate handling of the matter, and such persons must comply with any such directions.
- 9.3 If the COI Committee makes an initial determination that the alleged breach is not the result of an honest and reasonable error or oversight, then the COI Committee will normally investigate it pursuant to Section 9.4 of these Procedures.
- 9.4 The Chair of the COI Committee has the discretion to determine whether and how the COI Committee carries out its investigations, provided that such discretion must not be exercised in a manner that is inconsistent with the Policy or any associated Procedures or Rules or any directions issued by the Responsible Executive. Without limiting the generality of the foregoing, the Chair of the COI Committee may:
 - 9.4.1 determine that the investigation be terminated;
 - 9.4.2 recommend to the Responsible Executive that the investigation be undertaken as a Consolidated Investigation, as defined in UBC's Investigations Policy; and

- 9.4.3 request that the Responsible Executive appoint an external investigator or other person to undertake the investigation on behalf of the COI Committee.
- 9.5 Unless otherwise determined by the Chair of the COI Committee, investigations of complaints or concerns by the COI Committee will include:
 - 9.5.1 consulting with Central Human Resources, UBC administrators, and others, as appropriate;
 - 9.5.2 requesting written submissions or interviewing the Respondent and other parties with information that may be relevant to the alleged breaches, including the party who submitted the complaint or concern; and
 - 9.5.3 obtaining documents relevant to the alleged breach.
- 9.6 Upon completion of an investigation of the alleged breach, the COI Committee will prepare and deliver to the Responsible Executive a written report that includes the following information:
 - 9.6.1 a description of the alleged breach;
 - 9.6.2 the names of the parties who provided information and a summary of the information they provided;
 - 9.6.3 a summary of the relevant documents and other material reviewed;
 - 9.6.4 findings of fact based on the information, documents, and other material gathered during the investigation;
 - 9.6.5 a determination as to whether a breach of the Policy or any associated Procedures or Rules have occurred; and
 - 9.6.6 if a breach is determined to have occurred, then any mitigating or aggravating circumstances the COI Committee considers to be relevant.

10. Outcomes Resulting from Breaches

10.1 If, pursuant to Section 9.6 of these Procedures, the Responsible Executive receives a report from the COI Committee setting out a determination by the COI Committee that a breach of the Policy or any associated Procedures or Rules has occurred, the Responsible Executive will immediately forward the report to the Dean of the Respondent's faculty if the Respondent is a Faculty Member and to the Respondent's immediate supervisor if the Respondent is a Staff Member, with a copy to Central Human Resources. The Respondent's Dean or immediate supervisor will then consult with Central Human Resources to receive advice on what, if any, disciplinary action is appropriate. Any disciplinary action that is imposed on a UBC Person pursuant to the Policy or another UBC policy will be in accordance with the applicable terms and conditions of employment, including any applicable collective agreement, or such other contractual process(es) applicable to the UBC Person.

- 10.2 In addition to any disciplinary action, where a Faculty Member fails to comply with the Policy, the Dean or the Initial Reviewer (if the Initial Reviewer is not the Dean) may direct either the Office of Research Services or the University-Industry Liaison Office to freeze the research account(s) of the Faculty Member until further notice. UBC may also terminate access to funding or require the repayment of any financial benefit gained by the Faculty Member.
- 10.3 Failure to comply with the Policy and any associated Procedures and Rules could also lead to reporting by the Office of the Vice-President, Research and Innovation to the applicable Tri-Agency, or the U.S. Department of Health and Human Services, or any other funding and granting agencies and organizations and potentially result in the loss of research funding from such funding and granting agencies and organizations.

11. Establishing Guidelines and Rules

- 11.1 The Responsible Executive has the authority to establish, amend, and repeal Rules that address Conflicts of Commitment and Conflicts of Interest, provided that such Rules are not inconsistent with the Policy or any associated Procedures. The Responsible Executive must provide the Office of the University Counsel with a notice of the establishment, amendment, or repeal of any such Rules so that they can be reflected in the official repository of Board Policies maintained by the Office of the University Counsel pursuant to UBC's Regulatory Framework Policy. The establishment, amendment, or repeal of any Rules take effect and are binding on UBC Persons when approved by the Responsible Executive or on such later date as may be specified by the Responsible Executive.
- 11.2 The Responsible Executive may issue Guidelines as contemplated by UBC's Regulatory Framework Policy. Such Guidelines may address Conflict of Commitment or Conflict of Interest issues of interest to UBC Persons across all of UBC or to various subsets of UBC. Where an Administrative Head of Unit considers that Guidelines specific to their unit would be beneficial, the Administrative Head of Unit may recommend that the Responsible Executive issue such unit-specific Guidelines. The Responsible Executive must provide the Office of the University Counsel with a notice of the establishment, amendment, or repeal of any Guidelines so that they can be reflected in the official repository of Board Policies maintained by the Office of the University Counsel pursuant to UBC's Regulatory Framework Policy.
- 11.3 The Responsible Executive will normally consult with the COI Manager or the COI Committee, or both, when establishing, amending, or repealing any Rules or Guidelines.
- 11.4 Without limiting the generality of Section 11.1, the Rules may:
 - 11.4.1 provide for unit-specific processes for reviewing disclosures, taking into consideration the following:
 - (a) the volume of disclosures generated within the academic or administrative unit; and
 - (b) the need for specialized academic or professional knowledge and expertise to assess Conflicts of Interest or Conflicts of Commitment generated within the academic or administrative unit; and

11.4.2 establish conditional pre-approvals for defined Non-University Activities and defined categories of Non-University Activities that give rise to Conflicts of Interest or Conflicts of Commitment.

UBC	The University of British Columbia Board of Governors	Policy No.: SC3	
Long Title:	Conflict of Interest and Conflict of Commitment		
Short Title:	COI Policy		

Background & Purposes:

UBC's mission and mandate as a public institution of higher education is the advancement and dissemination of knowledge. UBC encourages all UBC Persons to seek and participate in research, to consult widely, and to engage in activities, including Non-University Activities, in a manner that does not interfere with their obligations to UBC.

Occasionally, UBC Persons may find themselves in either a Conflict of Interest or a Conflict of Commitment. The existence of a Conflict of Interest or Conflict of Commitment does not necessarily imply wrongdoing by a UBC Person. Such conflict(s) may exist regardless of a UBC Person's character, intentions, or motivations. However, Conflicts of Interest or Conflicts of Commitment that go unnoticed or are improperly managed may harm the reputation and integrity of the UBC Person(s) and UBC; and may also undermine the public's confidence in the abilities of the UBC Person(s) and UBC to pursue and disseminate knowledge, devoid of bias and personal interests. Without that public confidence, the effectiveness of UBC as a public institution and of UBC Persons is diminished.

The purposes of this Policy are to identify:

- situations that may give rise to a Conflict of Interest or Conflict of Commitment;
- a UBC Person's responsibility to identify, disclose, assess, and manage, if appropriate, a Conflict of Interest or Conflict of Commitment;
- the process for investigating any Conflict of Interest or Conflict of Commitment; and
- disciplinary actions that may result from a violation of this Policy or any associated Procedures or Rules.

Capitalized terms used in this Policy that are not otherwise defined herein shall have the meanings given to such terms in any associated Procedures or Rules.

1. General

1.1 All UBC Persons are responsible for informing themselves and complying with:

- 1.1.1 the requirements of this Policy and any associated Procedures and Rules; and
- 1.1.2 other applicable conflicts of interest or conflicts of commitment policies imposed by other third-party organizations, agencies, or institutions related to their activities at UBC.
- 1.2 All Faculty Members are responsible for informing themselves and complying with the requirements of any Tri-Agency or the U.S. Department of Health and Human Services; internal or external governance bodies (e.g., UBC's research ethics boards, UBC's Animal Care Committee, regulatory colleges such as the College of Physicians and Surgeons of British Columbia) that authorize research; other funding and granting agencies and organizations or third-party contracts, if related to their activities at UBC.
- 1.3 All UBC Persons are responsible for ensuring that participation in a UBC related activity and any Non-University Activity by a UBC Person is undertaken in compliance with this Policy, including ensuring such activities are conducted in a manner that:
 - 1.3.1 is consistent with UBC's mission and mandate to advance and disseminate knowledge;
 - 1.3.2 maintains the general public's trust and confidence that UBC and UBC Persons are able to undertake UBC's mission and mandate devoid of bias and personal interest; and
 - 1.3.3 is undertaken by UBC Persons acting with integrity and adhering to the highest ethical standards at all times, including declaring and appropriately managing any Conflict of Interest or Conflict of Commitment in a timely, open, forthright, constructive and accountable manner.

2. Conflicts of Commitment

2.1 Identifying Conflicts of Commitment

- 2.1.1 All UBC Persons are responsible for identifying, disclosing, and appropriately managing any Conflict of Commitment in accordance with this Policy and conducting themselves with the highest ethical standards, including considering any Conflict of Commitment in a manner that will bear the closest scrutiny.
- 2.1.2 UBC has not attempted to develop an exhaustive list of Conflicts of Commitment as each situation is dependent upon its specific facts. However, the following are examples of situations where a Conflict of Commitment is deemed to exist:
 - (a) where a UBC Person engages in a Non-University Activity during their normal work hours at UBC;
 - (b) where a UBC Person employed on a full-time basis by UBC enrolls in adegree or diploma program at UBC, or any other educational institution;
 - (c) where a UBC Person uses UBC resources for Non-University Activities;

- (d) where a full-time Faculty Member's total involvement in Outside Professional Activities is greater than 52 days per year, including evenings, weekends and vacations; or
- (e) where a full-time Faculty Member will be away from their place of work at UBC for a period of 30 consecutive days (excluding holidays) as a result of the Faculty Member's Outside Professional Activities.

2.2 **Disclosing Conflicts of Commitment**

- 2.2.1 If a UBC Person intends to engage in any activity, including any Non-University Activities, that could lead to a Conflict of Commitment, then the UBC Person must immediately disclose that activity to their Initial Reviewer pursuant to Section 3 of the Procedures. These disclosure obligations <u>also</u> apply to all UBC Persons as soon as <u>knowledge of</u> the Conflict of Commitment arises and continue until the Conflict of Commitment ceases to exist.
- 2.2.2 In addition, all Faculty Members and Staff Members are required to file annual Conflict of Commitment disclosures.
- 2.2.3 If a Faculty Member intends to engage in any activity that exceeds the thresholds set out in Section 2.1.2(d) or Section 2.1.2(e) of this Policy, then they must disclose and obtain prior written authorization from the Dean of the faculty in which they hold their primary appointment, a copy of which must be filed with the COI Manager.
- 2.2.4 All UBC Persons are encouraged to bring any questions or concerns regarding their Conflicts of Commitment to their Initial Reviewer, who may consult with the COI Manager.

2.3 Assessing and Managing Conflicts of Commitment

- 2.3.1 If an Initial Reviewer becomes aware of a Conflict of Commitment, whether through disclosure or otherwise, then they must assess the Conflict of Commitment taking into consideration the following:
 - (a) whether a UBC Person's engagement in the Non-University Activity interferes with their professional obligations to UBC, including the elements set forth in Section 1.3 of this Policy; and
 - (b) whether the benefit to UBC from a UBC Person's engagement in the Non-University Activity outweighs the interference with their professional obligations to UBC or would outweigh the interference if appropriately managed through a Management Plan.
- 2.3.2 Once the Initial Reviewer has assessed the Conflict of Commitment situation, the Initial Reviewer, may determine what action, if any, is appropriate pursuant to Section 4 of the Procedures.

3. Conflicts of Interest

3.1 Identifying Conflicts of Interest

- 3.1.1 All UBC Persons are responsible for identifying, disclosing, and appropriately managing any Conflict of Interest in accordance with this Policy and conducting themselves at all times with the highest ethical standards, including considering any Conflict of Interest in a manner that will bear the closest scrutiny.
- 3.1.2 Any UBC Person who engages in Non-University Activities is responsible for identifying any Conflict of Interests arising naturally from their engagement in Non-University Activities.
- 3.1.3 Identifying a Conflict of Interest involves two elements:
 - (a) the UBC Person's or a Related Party's personal interest(s), including without limitation, business, commercial or Financial Interests and personal matters and career interests; and
 - (b) the UBC Person's professional obligations to UBC.
- 3.1.4 A Conflict of Interest includes the following situations:
 - (a) an actual Conflict of Interest;
 - (b) a situation where the UBC Person's personal interest has not yet conflicted but has the potential or might be expected to conflict with the UBC Person's professional obligations to UBC; or
 - (c) the perception of a Conflict of Interest from the perspective of a reasonably wellinformed, impartial observer.
- 3.1.5 UBC has not attempted to develop an exhaustive list of Conflicts of Interest as each situation is dependent upon its specific facts. However, a non-exhaustive list of situations where a Conflict of Interest is deemed to exist is set forth in Section 2 of the Procedures.

3.2 Disclosing Conflicts of Interest

3.2.1 If a UBC Person intends to engage in an activity that could lead to a Conflict of Interest, then the UBC Person must immediately disclose that activity to their Initial Reviewer pursuant to Section 3 of the Procedures and to all parties affected by the Conflict of Interest. For example, if a UBC Person makes a public statement, in the media or otherwise, and identifies themselves as being associated with UBC, then the UBC Person must disclose, as part of that public statement, any other relevant relationships, circumstances, or engagements in other Non-University Activities that give rise to a Conflict of Interest.

- 3.2.2 If prior disclosure is impossible, then disclosure must be made as soon as possible after commencement of the activity.
- 3.2.3 Prior to commencing a research project, a UBC Person who is a principal investigator or researcher must disclose all Conflicts of Interest that may arise in relation to the research project.
- 3.2.4 These disclosure obligations <u>also</u> apply to all UBC Persons as soon as <u>knowledge of</u> the Conflict of Interest arises and continue until the <u>Conflict</u> of Interest ceases to exist.
- 3.2.5 In addition, all Faculty Members and Staff Members are required to file annual Conflict of Interest disclosures.
- 3.2.6 All UBC Persons are encouraged to bring any questions or concerns regarding their Conflicts of Interest to their Initial Reviewer, who may consult with the COI Manager.

3.3 Assessing and Managing Conflicts of Interest

- 3.3.1 If an Initial Reviewer becomes aware of a Conflict of Interest, whether through disclosure or otherwise, then they must assess the situation to determine what action, if any, is appropriate, including taking into consideration the following:
 - (a) the impact on the UBC Person's ability to satisfy their professional obligations to UBC, including the elements set forth in Section 1.3 of this Policy;
 - (b) the degree to which the proposed action will be detrimental to the interests of UBC;
 - (c) the degree to which it may harm the reputation and integrity of the UBC Person(s) or UBC, or both;
 - (d) the degree to which it may compromise the professional judgment of any UBC Person involved in conducting or reporting research;
 - (e) the extent to which the proposed action or activity could be managed through an appropriate Management Plan; and
 - (f) any of the other Conflict of Interest elements set forth in Section 2 of the Procedures.
- 3.3.2 Once the Initial Reviewer has assessed the Conflict of Interest situation, the Initial Reviewer, may determine what action, if any, is appropriate pursuant to Section 4 of the Procedures.
- 4. Reporting of Complaints or Concerns

- 4.1 All persons are encouraged to contact the COI Manager in writing if they believe that a Conflict of Interest or Conflict of Commitment has been unreported, misrepresented, unassessed, or unmanaged or that there has been any other breach of this Policy or any associated Procedures or Rules. Upon receipt, the COI Manager will handle the complaint or concern in accordance with Section 6.1.4 of the Procedures.
- 4.2 All complaints or concerns submitted pursuant to this Policy will be taken seriously. The anonymity of the person making the complaint or raising a concern will, unless otherwise directed by that person, be maintained, except as required by a legal process or the law. However, any information provided anonymously cannot be acted upon unless the information is independently substantiated. Allegations of retaliation against any person who, in good faith, makes a complaint or raises a concern pursuant to this Policy, gives evidence, or otherwise participates in a process under this Policy will be handled in accordance with UBC's Retaliation Policy.
- 4.3 UBC will protect the personal information of all parties involved as required under applicable legislation, including the *Freedom of Information and Protection of Privacy Act* (British Columbia). Except as required by a legal process or law, any <u>personal</u> information disclosed by a UBC Person pursuant to this Policy will be held in confidence and will only be available to relevant granting or funding agencies and those with a need to know the information in order to fulfill their responsibilities to UBC, including:
 - 4.3.1 those responsible under this Policy and any associated Procedures or Rules for the review, management, or audit of Conflict of Interest and Conflict of Commitment disclosures; or
 - 4.3.2 those who are involved in investigations of alleged breaches of UBC Policies or other misconduct, which may include: Central Human Resources; the Office of the Vice President, Research and Innovation; the Office of the University Counsel; members of the COI Committee; the Responsible Executive; or any of their respective and applicable delegates.

UBC

PROCEDURES ASSOCIATED WITH THE POLICY

Pursuant to the Regulatory Framework Policy, the President may approve Procedures or the amendment or repeal of Procedures. Such approvals must be reported at the next meeting of the UBC Board of Governors or as soon thereafter as practicable.

Capitalized terms used in the Policy that are not otherwise defined herein shall have the meanings given to such terms in these Procedures or Rules.

1. **Definitions**

- 1.1 "Administrative Head(s) of Unit" means a Director of an administrative unit; a Head of an academic department; a Director of a centre, institute or school; a Principal of a college; a Dean of a faculty; the Registrar; the University Librarian; a Vice-President; or the President or the equivalent; or any of their respective delegates, which may include Associate Vice-Presidents; Vice-Principals; and Associate Vice-Principals.
- 1.2 "**Business**" means a corporation, partnership, sole proprietorship, firm, franchise, association, organization, holding company, joint stock company, receivership, business or real estate trust, or other entity organized for profit, not for profit, or charitable purposes, but excluding a UBC affiliated Hospital or a medical practice or other entity controlled by UBC or an affiliated Hospital.
- 1.3 "Central Human Resources" means Employee and Labour Relations or Faculty Relations on the UBC Vancouver campus, or Human Resources on the UBC Okanagan campus.
- 1.4 **"COI Committee"** means a committee established by the Responsible Executive to carry out the roles and responsibilities set out in the Policy, and which includes at least one independent person who is not a UBC Person.
- 1.5 "**COI Manager**" means the individual appointed by the Responsible Executive to oversee the administration and governance of the Policy or other duties as delegated by the Responsible Executive from time to time, and, whose contact information can be found on the website of the office of the Responsible Executive.
- 1.6 "Conflict(s) of Commitment" means a situation where a UBC Person engages in Non-University Activities that adversely affect the discharge of the UBC Person's primary responsibilities to UBC.
- 1.7 "**Conflict(s) of Interest**" means a situation in which a UBC Person, or their Related Party(ies), has a personal interest that actually conflicts, could conflict, or there is a perception that it conflicts with the UBC Person's obligations to UBC.
- 1.8 **"Faculty Member**" means an individual appointed by the University in the following faculty titles, ranks and descriptions: professor, associate professor, assistant professor, professor of

teaching, assistant professor of teaching, associate professor of teaching, adjunct professor, clinical professor, clinical associate professor, clinical assistant professor, clinical instructor, clinical fellow, academic program director, librarian, research associate, post-doctoral fellow, sessional lecturer, lecturer, visiting assistant professor, visiting associate professor, visiting professor, visiting lecturer, visiting scientist, visiting scholar, emeritus, honorary faculty or affiliate appointment or in an equivalent position designated by the applicable senate.

- 1.9 **"Financial Interest"** means having or having the expectation to receive:
 - 1.9.1 greater than a 5% ownership interest in a Business; or
 - 1.9.2 anything with a monetary value exceeding \$5,000 in any one calendar year, including remuneration (e.g., salary, consulting fees, retainers, honoraria, bonuses, gifts, speaker's fees, advisory board remuneration, finders or recruitment fees), equity interests (e.g., stocks, stock options or other ownership interests), and intellectual property rights (e.g., patents, copyrights, royalties or other payments from such rights).
- 1.10 "Guidelines" has the meaning set forth in UBC's Regulatory Framework Policy.
- 1.11 "Initial Reviewer(s)" means the individual responsible for reviewing a UBC Person's disclosures of Conflicts of Interest or Conflicts of Commitment. For Faculty Members, the Initial Reviewer is normally the Dean of the faculty in which they hold their primary appointment, unless their faculty is departmentalized, in which case it is normally the Head of their department or school. For Staff Members, the Initial Reviewer is normally their immediate supervisor and, where the Staff Member has more than one immediate supervisor, the Initial Reviewer is the immediate supervisor designated by the Responsible Executive. For all other UBC Persons, the Initial Reviewer is normally their immediate supervisor. Notwithstanding the foregoing, the Responsible Executive may designate a different Initial Reviewer for any UBC Persons.
- 1.12 **"Management Plan**" means the written document governing and managing a UBC Person's participation in an activity deemed to constitute a Conflict of Interest or Conflict of Commitment.
- 1.13 "Non-University Activity(ies)" means any activity outside a UBC Person's scope of work with UBC and includes Outside Professional Activities.
- 1.14 "Outside Professional Activity(ies)" means any activity outside a UBC Person's scope of work with UBC that involves the same specialized skill and knowledge that the UBC Person utilizes in their work with UBC and includes the operation of a Business, consulting or advisory services, external teaching, external academic appointments and external speaking engagements.
- 1.15 "**Policy**" means the COI Policy.
- 1.16 **"Position of Authority"** means any position that includes responsibilities for a material function of the operation or management of a Business.
- 1.17 "Related Party(ies)" means a UBC Person's immediate family member (e.g., spouse, partner, or sibling), direct descendant or ancestor (e.g., child, grandchild, parent, or grandparent), any other person living in the same household, any other person with whom the UBC Person shares

a Financial Interest, either directly or indirectly, or any Business in which the UBC Person has an ownership interest of more than 5%.

- 1.18 "**Related Policy(ies)**" means the related policies specific to the Policy as identified in the official repository for the Policy maintained by the Office of the University Counsel and published on its website.
- 1.19 "**Respondent**" means the UBC Person who is the subject of an investigation of an alleged breach of the COI Policy or associated Procedures or Rules.
- 1.20 **"Responsible Executive**" means individual(s) assigned by the President, pursuant to UBC's Regulatory Framework Policy, to be responsible for the Policy; and any sub-delegate of that assigned responsible individual(s) except to the extent that the power to delegate is specifically excluded in the Policy or in the appointment by the President.
- 1.21 "Rules" has the meaning set forth in UBC's Regulatory Framework Policy.
- 1.22 "Staff Member" means individual(s) employed by UBC in any of the following employee categories or titles:
 - 1.22.1 Management and Professional;
 - 1.22.2 Director of a service unit;
 - 1.22.3 Administrative AVP (as defined in UBC's Administrative AVPs Policy);
 - 1.22.4 Administrative VP (as defined in UBC's Administrative VPs Policy); and
 - 1.22.5 Any other UBC Person(s), who is not a Faculty Member, as required by the Responsible Executive (e.g., those who have signing authority over UBC financial accounts).
- 1.23 **"Tri-Agency**" means the Canadian Institutes of Health Research, the Natural Sciences and Engineering Research Council, and the Social Sciences and Humanities Research Council, collectively.
- 1.24 "UBC Person(s)" means all Faculty Members, all Staff Members, all other employees of UBC, all students to the extent that they are carrying out UBC research, and all individual(s) who have agreed to carry out duties on behalf of UBC as a volunteer.

2. Additional Conflict of Interest Considerations:

- 2.1 The following is a non-exhaustive list of situations where <u>Conflictsa Conflict</u> of Interest is deemed to exist:
 - 2.1.1 where a UBC Person's responsibility to instruct, evaluate and supervise students in a fair, unbiased and effective manner is or could be impeded or compromised (Note, the inherent power imbalance that exists between a UBC Person and a student must not be used for personal benefit.);

- 2.1.2 where a UBC Person or a Related Party has a Financial Interest in the outcome of their research, including any Financial Interest in a company sponsoring research at their UBC laboratory or facilities;
- 2.1.3 where a UBC Person or a Related Party has a Financial Interest in their teaching activities at UBC, other than their annual salary from UBC;
- 2.1.4 where a UBC Person has influence over a decision about a proposed relationship between UBC and a Business in which the UBC Person or a Related Party has a Financial Interest or holds a Position of Authority;
- 2.1.5 where a UBC Person or a Related Party has or obtains a Financial Interest or a Position of Authority in a Business with which UBC has an existing relationship and the Business is related to the UBC Person's work at UBC;
- 2.1.6 where a UBC Person transfers research results or intellectual property arising from their UBC research to a Business in which the UBC Person or a Related Party has a Financial Interest or holds a Position of Authority;
- 2.1.7 where a UBC Person is in a position to influence human resource decisions (such as recruitment, offer of employment, <u>appointment or selection process</u>, evaluation of performance, promotion, granting of tenure, <u>discipline</u>, <u>investigations</u>, or termination of employment) or admission decisions with respect to a person who is a Related Party or who has another relationship with the UBC Person that might reasonably be perceived as creating a Conflict of Interest;
- 2.1.8 where a UBC Person uses or attempts to use their position with UBC to solicit students, other UBC Persons, government agencies, private companies, or members of the public for Non-University Activities, including employing students or other UBC Persons to perform services for the UBC Person or a Related Party on personal or commercial matters;
- 2.1.9 where a UBC Person uses information that is acquired as a result of their position with UBC and not in the public domain for Non-University Activities, unless the UBC Person has proprietary rights (usually enforceable through copyright) to that information (Note, UBC Persons should also be aware that insider trading restrictions may also apply to them under the laws of British Columbia or other jurisdictions.);
- 2.1.10 where a UBC Person's obligations to a board of directors, advisory boards, or the like of an outside organization interfere with or compromise the UBC Person's obligations to UBC;
- 2.1.11 where a UBC Person accepts gifts with a value of \$500 or more that would not have been offered if not for the UBC Person's position or role at UBC;
- 2.1.12 where a Faculty Member enrolls as a student in their own faculty; or

2.1.13 where personal considerations could be reasonably perceived to compromise a UBC Person's professional judgment in conducting or reporting research, teaching, or carrying out administrative activities at UBC.

3. Required Disclosures

- 3.1 All UBC Persons must:
 - 3.1.1 immediately disclose to their Initial Reviewer if a UBC Person intends to engage in any Non-University Activities that could lead to a Conflict of Interest and Conflict of Commitment:
 - <u>3.1.2</u> <u>3.1.1</u> immediately disclose Conflicts of Interest and Conflicts of Commitment to their Initial Reviewer as soon as such situation or event arises; and
 - <u>3.1.3</u> <u>3.1.2</u> continue to disclose and submit such information, as requested, or at least annually until the Conflicts of Interest and Conflicts of Commitment cease to exist.
- 3.2 At least annually, all Faculty Members <u>and Staff Members</u> must submit a completed disclosure that discloses the following:
 - 3.2.1 all Outside Professional Non-University Activities;
 - 3.2.2 all circumstances (if any) that give rise to a Conflict of Interest or Conflict of Commitment; and
 - 3.2.3 such additional information asked for in the prescribed annual disclosure form, as it may be amended from time to time.
- 3.3 At least annually, all Staff Members must submit a completed disclosure that discloses the following:
 - 3.3.1 all Non-University Activities, if any;
 - 3.3.2 all circumstances (if any) that give rise to a Conflict of Interest or Conflict of Commitment;
 - 3.3.3 such additional information asked for in the prescribed annual disclosure form, as it may be amended from time to time.
- 3.3 3.4 All disclosures made by Faculty Members under Sections 3.1 and Section 3.2 of these Procedures must be submitted utilizing the Conflict of Interest or the Conflict of Commitment module in the *RISe* system.
- <u>3.4</u> <u>3.5</u> All disclosures made by Staff Members under <u>Sections 3.1 and 3.3</u><u>Section 3.2</u> of these Procedures must be submitted utilizing the Conflict of Interest or the Conflict of Commitment module in the *Workday* system.
- 3.5 3.6-All disclosures made by UBC Persons under Section 3.1 of these Procedures who are not Faculty Members or Staff Members must be submitted utilizing the Conflict of Interest or Conflict of Commitment module in the system deemed most appropriate by the Initial Reviewer.

4. Role of Initial Reviewer

- 4.1 The Initial Reviewer is responsible for assessing the disclosure and determining whether there are any Conflicts of Interest or Conflicts of Commitment, and, if so, whether they can and should be permitted subject to a Management Plan.
- 4.2 If the Initial Reviewer determines that a proposed action or activity, including a proposed research project, is:
 - 4.2.1 permissible as disclosed with no further action, then the Initial Reviewer's determination will be recorded in writing and the UBC Person may proceed with the action or activity subject to obtaining all other applicable approvals;
 - 4.2.2 permissible only if appropriately managed through the implementation of a Management Plan, then the Initial Reviewer will develop and implement a Management Plan to manage the Conflict of Interest or the Conflict of Commitment and the UBC Person may proceed with the action or activity pursuant to the Management Plan; or
 - 4.2.3 not permissible, then the UBC Person must not proceed with the proposed action or activity giving rise to the Conflict of Interest or Conflict of Commitment.
- 4.3 The Initial Reviewer may consult with the COI Manager in relation to any decisions required to be made under Section 4 of these Procedures, including the development of any applicable Management Plan(s) and the COI Manager may in turn consult with the COI Committee.
- 4.4 If the Initial Reviewer is unable to determine whether a Conflict of Interest or a Conflict of Commitment or any proposed action or activity is permissible due to their own Conflict of Interest or otherwise, then the Initial Reviewer must refer the disclosure one level up for handling. If such a referral is not appropriate, then the matter will be referred to the Responsible Executive for a final determination of who should handle the matter.
- 4.5 Where the UBC Person is a Faculty Member who holds an appointment in more than one faculty, the Initial Reviewer must keep the Dean of the other faculty up-to-date with a copy of that Faculty Member's disclosures and any applicable Management Plan(s). Where the UBC Person is a Staff Member who has more than one immediate supervisor, then the Initial Reviewer must consult with and keep the other immediate supervisors up-to-date with a copy of that Staff Member's disclosures and any applicable Management Plan(s).

5. Unit-Specific Review Process

5.1 Notwithstanding Section 4 of these Procedures, if the UBC Person disclosing a Conflict of Interest or a Conflict of Commitment is part of an academic or administrative unit for which unit-specific review processes have been established pursuant to one or more Rules, then the review shall be undertaken in accordance with such Rules.

6. Role of the COI Manager

- 6.1 The COI Manager is charged with the following responsibilities:
 - 6.1.1 serving as the secretary to the COI Committee;
 - 6.1.2 providing advice and guidance to Initial Reviewers, Administrative Heads of Unit, the Responsible Executive, and the COI Committee with respect to any matter pertaining to the Policy and any associated Procedures and Rules, including without limitation:
 - (a) the establishment of processes for the assessment of Conflicts of Interest and Conflicts of Commitment;
 - (b) the assessment of disclosures of Conflicts of Interest and Conflicts of Commitment; and
 - (c) the development of appropriate Management Plan(s);
 - 6.1.3 making recommendations to the Responsible Executive in relation to the issuance of any Rules, Guidelines, templates, or other materials or resources and consulting the COI Committee as appropriate;
 - 6.1.4 handling any complaints or concerns received pursuant to Section 4.1 of the Policy as the COI Manager deems appropriate, including forwarding such complaints or concerns:
 - (a) to the appropriate Initial Reviewer or any other person identified in a Rule establishing a unit-specific review process contemplated by Section 5 of these Procedures for assessment and management pursuant to either Section 2.3 or Section 3.3 the Policy; or
 - (b) to the attention of the COI Committee for handling pursuant to Section 9 of these Procedures; and
 - 6.1.5 providing training to UBC Persons regarding Conflicts of Interest, Conflicts of Commitment, and the application of the Policy and any associated Procedures and Rules.

7. Role of the COI Committee

- 7.1 The Responsible Executive must establish a COI Committee.
- 7.2 The COI Committee is charged with the following responsibilities:
 - 7.2.1 auditing random samples of Conflicts of Interest and Conflicts of Commitment that have been approved and any related Management Plans for quality control purposes and advising Initial Reviewers on appropriate standards for the quality of disclosures, reviews, and Management Plans;

- 7.2.2 acting as a sounding board and source of advice regarding the appropriate handling of specific disclosures or other cases referred to it by the COI Manager.
- 7.2.3 serving as the final avenue of appeal (such as where a UBC Person disagrees with the assessment of an Initial Reviewer or the assessment resulting from a Rule established under Section 11 of these Procedures);
- 7.2.4 to review concerns or issues in accordance with Section 9 of these Procedures, whether self-initiated or brought forth by the COI Manager pursuant to Section 6.1.4(b) of these Procedures; and
- 7.2.5 fulfilling such other roles and responsibilities as may be assigned to it by the Responsible Executive.

8. Appeals

- 8.1 All UBC Persons may appeal any decision regarding a Conflict of Interest or Conflict of Commitment made by their Initial Reviewer in accordance with Section 4 of these Procedures by sending a notice of appeal to the COI Manager in writing using such form as may be prescribed by the COI Manager. The COI Manager will bring the notice to the attention of the COI Committee.
- 8.2 The COI Committee may, at its discretion, refuse to hear an appeal that is filed more than 30 days after the date of the Initial Reviewer's decision.
- 8.3 The Chair of the COI Committee has the discretion to determine the manner in which the COI Committee considers appeals, provided that such discretion must not be exercised in a manner that is inconsistent with the Policy or any associated Procedures or Rules or any directions issued by the Responsible Executive.

9. Referral of Complaints or Concerns to the COI Committee

- 9.1 If a complaint or concern is forwarded to the attention of the COI Committee by the COI Manager pursuant to Section 6.1.4(b) of these Procedures, the COI Committee will normally make an initial determination as to whether the alleged breach has resulted from an honest and reasonable error or oversight. The COI Committee may reconsider its initial determination at any time.
- 9.2 Where the COI Committee's initial determination is that the alleged breach has resulted from an honest and reasonable error or oversight, the COI Committee will normally advise or direct the COI Manager, the Respondent, their Initial Reviewer and any other person that the COI Committee determines should receive such advice or directions as to the appropriate handling of the matter, and such persons must comply with any such directions.
- 9.3 If the COI Committee makes an initial determination that the alleged breach is not the result of an honest and reasonable error or oversight, then the COI Committee will normally investigate it pursuant to Section 9.4 of these Procedures.

- 9.4 The Chair of the COI Committee has the discretion to determine whether and how the COI Committee carries out its investigations, provided that such discretion must not be exercised in a manner that is inconsistent with the Policy or any associated Procedures or Rules or any directions issued by the Responsible Executive. Without limiting the generality of the foregoing, the Chair of the COI Committee may:
 - 9.4.1 determine that the investigation be terminated;
 - 9.4.2 recommend to the Responsible Executive that the investigation be undertaken as a Consolidated Investigation, as defined in UBC's Investigations Policy; and
 - 9.4.3 request that the Responsible Executive appoint an external investigator or other person to undertake the investigation on behalf of the COI Committee.
- 9.5 Unless otherwise determined by the Chair of the COI Committee, investigations of complaints or concerns by the COI Committee will include:
 - 9.5.1 consulting with Central Human Resources, UBC administrators, and others, as appropriate;
 - 9.5.2 requesting written submissions or interviewing the Respondent and other parties with information that may be relevant to the alleged breaches, including the party who submitted the complaint or concern; and
 - 9.5.3 obtaining documents relevant to the alleged breach.
- 9.6 Upon completion of an investigation of the alleged breach, the COI Committee will prepare and deliver to the Responsible Executive a written report that includes the following information:
 - 9.6.1 a description of the alleged breach;
 - 9.6.2 the names of the parties who provided information and a summary of the information they provided;
 - 9.6.3 a summary of the relevant documents and other material reviewed;
 - 9.6.4 findings of fact based on the information, documents, and other material gathered during the investigation;
 - 9.6.5 a determination as to whether a breach of the Policy or any associated Procedures or Rules have occurred; and
 - 9.6.6 if a breach is determined to have occurred, then any mitigating or aggravating circumstances the COI Committee considers to be relevant.

10. Outcomes Resulting from Breaches

- 10.1 If, pursuant to Section 9.6 of these Procedures, the Responsible Executive receives a report from the COI Committee setting out a determination by the COI Committee that a breach of the Policy or any associated Procedures or Rules has occurred, the Responsible Executive will immediately forward the report to the Dean of the Respondent's faculty if the Respondent is a Faculty Member and to the Respondent's immediate supervisor if the Respondent is a Staff Member, with a copy to Central Human Resources. The Respondent's Dean or immediate supervisor will then consult with Central Human Resources to receive advice on what, if any, disciplinary action is appropriate. Any discipline will be commensurate with the seriousness of the violation. Any disciplinary action that is imposed on a UBC Person pursuant to the Policy or another UBC policy will be in accordance with the applicable terms and conditions of employment, including any applicable collective agreement, or such other contractual process(es) applicable to the UBC Person.
- 10.2 In addition to any disciplinary action, where a Faculty Member fails to comply with the Policy, the Dean or the Initial Reviewer (if the Initial Reviewer is not the Dean) may direct <u>either</u> the Office of Research Services <u>or the University-Industry Liaison Office</u> to freeze the research account(s) of the Faculty Member until further notice. UBC may also terminate <u>access to</u> funding or require the repayment of any financial benefit gained by the Faculty Member.
- 10.3 Failure to comply with the Policy and any associated Procedures and Rules could also lead to reporting by the Office of the Vice-President, Research and Innovation to the applicable Tri-Agency, or the U.S. Department of Health and Human Services, or any other funding and granting agencies and organizations and potentially result in the loss of research funding from such funding and granting agencies and organizations.

11. Establishing Guidelines and Rules

- 11.1 The Responsible Executive has the authority to establish, amend, and repeal Rules that address Conflicts of Commitment and Conflicts of Interest, provided that such Rules are not inconsistent with the Policy or any associated Procedures. The Responsible Executive must provide the Office of the University Counsel with a notice of the establishment, amendment, or repeal of any such Rules so that they can be reflected in the official repository of Board Policies maintained by the Office of the University Counsel pursuant to UBC's Regulatory Framework Policy. The establishment, amendment, or repeal of any Rules take effect and are binding on UBC Persons when approved by the Responsible Executive or on such later date as may be specified by the Responsible Executive.
- 11.2 The Responsible Executive may issue Guidelines as contemplated by UBC's Regulatory Framework Policy. Such Guidelines may address Conflict of Commitment or Conflict of Interest issues of interest to UBC Persons across all of UBC or to various subsets of UBC. Where an Administrative Head of Unit considers that Guidelines specific to their unit would be beneficial, the Administrative Head of Unit may recommend that the Responsible Executive issue such unit-specific Guidelines. The Responsible Executive must provide the Office of the University Counsel with a notice of the establishment, amendment, or repeal of any Guidelines so that they can be reflected in the official repository of Board Policies maintained by the Office of the University Counsel pursuant to UBC's Regulatory Framework Policy.

- 11.3 The Responsible Executive will normally consult with the COI Manager or the COI Committee, or both, when establishing, amending, or repealing any Rules or Guidelines.
- 11.4 Without limiting the generality of Section 11.1, the Rules may:
 - 11.4.1 provide for unit-specific processes for reviewing disclosures, taking into consideration the following:
 - (a) the volume of disclosures generated within the academic or administrative unit; and
 - (b) the need for specialized academic or professional knowledge and expertise to assess Conflicts of Interest or Conflicts of Commitment generated within the academic or administrative unit; and
 - 11.4.2 establish conditional pre-approvals for defined Non-University Activities and defined categories of Non-University Activities that give rise to Conflicts of Interest or Conflicts of Commitment.

Summary report:			
Litera Compare for Word 11.7.0.54 Document comparison done on 2/13/2024 2:51:40 PM			
Style name: Default Style			
Intelligent Table Comparison: Active			
Original filename: Proposed Amended COI Policy (Nov 2023	B).docx		
Modified filename: Proposed COI Policy and Procedures being presented for			
Board of Governors' approval.docx			
Changes:			
Add	27		
Delete	26		
Move From	0		
Move To	0		
Table Insert	0		
Table Delete	0		
Table moves to	0		
Table moves from	0		
Embedded Graphics (Visio, ChemDraw, Images etc.)	0		
Embedded Excel	0		
Format changes	0		
Total Changes:	53		

UBC	The University of British Columbia Board of Governors	Policy No.: SC3
Long Title:	Conflict of Interest and Conflict of Commitr	nent
Short Title	: COI Policy	

Background & Purposes:

The University UBC's mission and mandate as a public institution of higher education is responsible for the advancement and dissemination of knowledge. As such, itUBC encourages all UBC Persons to seek and participate in research, to consult widely, and to engage in activities, including Non-University Activities so long as such activities do, in a manner that does not interfere with their obligations to the University. While the University recognizes that these activities may benefit the participants, the University, and the public at large, the University is committed to ensuring that these activities are conducted in a manner consistent with the interests and mission of the University and in a manner that maintains the community's trust and confidence. Faculty Members, staff and students of the University must act with integrity and adhere to the highest ethical standards at all times UBC.

Occasionally and as a result of normal and productive engagements inside and outside the University, Faculty Members, staff and students may find themselves in a conflict of commitment, an actual or potential conflict of interest, or in a situation where there is a perception of a conflict of interest. In fact, many beneficial and productive developments and initiatives, such as the University's commitment to liaise with industry and to transfer technology, the growth of industry-sponsored research, and the substantial increase in the University's interactions with outside organizations, also generally lead to an increase in conflicts of commitment, conflicts of interest and perceived conflicts of interest.

Conflicts of commitment, conflicts of interest and perceived conflicts of interest, UBC Persons may find themselves in either a Conflict of Interest or a Conflict of Commitment. The existence of a Conflict of Interest or Conflict of Commitment does not necessarily imply wrongdoing by a UBC Person. Such conflict(s) may exist regardless of a UBC Person's character, intentions, or motivations. However, Conflicts of Interest or Conflicts of Commitment that go unnoticed or are improperly managed threaten to impugnmay harm the reputation and integrity of the persons involved and, potentially, the University as a whole. TheyUBC Person(s) and UBC; and may also undermine the public's confidence in the University's andabilities of the UBC Person'(s-ability) and UBC to pursue and disseminate knowledge, devoid of bias and personal interests. Without that public confidence, the effectiveness of the UniversityUBC as a public institution and of UBC Persons-as intellectual leaders is diminished. Moreover, UBC Persons are expected to vigilantly guard against conflicts of commitment, actual and potential conflicts of interest, and perceived conflicts of interest. It is important to note that the mere existence of a conflict of commitment, a conflict of interest or a perceived conflict of interest does not necessarily imply wrongdoing on anyone's part. Rather, conflicts exist regardless of a person's character, intentions and motivations. In situations where a conflict exists or there is a perception of a conflict, it is not sufficient to merely live up to one's obligations; it is necessary to do so

The purposes of this Policy are to identify:

- situations that may give rise to a Conflict of Interest or Conflict of Commitment;
- <u>a UBC Person's responsibility to identify, disclose, assess, and manage, if appropriate, a Conflict</u> of Interest or Conflict of Commitment;
- the process for investigating any Conflict of Interest or Conflict of Commitment; and
- <u>disciplinary actions that may result from a violation of this Policy or any associated Procedures</u> or Rules.

Capitalized terms used in this Policy that are not otherwise defined herein shall have the meanings given to such terms in any associated Procedures or Rules.

1. General

- 1.1 All UBC Persons are responsible for informing themselves and complying with:
 - 1.1.1 the requirements of this Policy and any associated Procedures and Rules; and
 - 1.1.2 <u>other applicable conflicts of interest or conflicts of commitment policies imposed by</u> <u>other third-party organizations, agencies, or institutions related to their activities at</u> <u>UBC.</u>
- 1.2 All Faculty Members are responsible for informing themselves and complying with the requirements of any Tri-Agency or the U.S. Department of Health and Human Services; internal or external governance bodies (e.g., UBC's research ethics boards, UBC's Animal Care Committee, regulatory colleges such as the College of Physicians and Surgeons of British Columbia) that authorize research; other funding and granting agencies and organizations or third-party contracts, if related to their activities at UBC.
- 1.3 <u>All UBC Persons are responsible for ensuring that participation in a UBC related activity and any</u> <u>Non-University Activity by a UBC Person is undertaken in compliance with this Policy,</u> <u>including ensuring such activities are conducted in a manner that:</u>
 - 1.3.1 is consistent with UBC's mission and mandate to advance and disseminate knowledge;

- 1.3.2 maintains the general public's trust and confidence that UBC and UBC Persons are able to undertake UBC's mission and mandate devoid of bias and personal interest; and
- 1.3.3 is undertaken by UBC Persons acting with integrity and adhering to the highest ethical standards at all times, including declaring and appropriately managing any Conflict of Interest or Conflict of Commitment in a timely, open, forthright, constructive and accountable manner.

Therefore, the purpose of this Policy is to: (a) identify conflicts of commitment, actual and potential conflicts of interest, and situations that may give rise to the perception of a conflict of interest; and (b) to provide procedures whereby such situations are disclosed and either avoided or appropriately managed. Management, in some circumstances, may mean avoiding the activity; and in others it may mean implementing a management protocol.

This process is not only important to preserve the integrity of the persons and processes involved, but also to ensure that the University and UBC Persons remain compliant with the requirements of the various agencies that provide research and other funding. In particular, requirements from the Tri-Council agencies, the U.S. Department of Health and Human Services, and other bodies are in place, which require the University to create and implement policies and systems to identify and manage conflicts of interest in specific ways (for example, the requirement to file an annual conflict of interest report is a requirement of the Tri-Council agencies). Failure to comply with these requirements could lead to the loss of a substantial source of research funding.

Identifying and managing conflicts of interest and perceived conflicts of interest is relevant to all research subject matters, including but not limited to research involving human subjects, and to the administration of the University.

Accordingly, all UBC Persons are responsible for informing themselves and complying with requirements imposed on them by this Policy, all related policies (the Human Research Policy), and the requirements of all agencies and organizations from which they receive funding.

Defined terms are capitalized in this Policy and can be found in Section 8 at the end of this Policy.

1. Scope

- 1.1 This Policy applies to all UBC Persons, which means full-time and part-time Faculty Members and staff members of the University, and any other person who teaches, conducts research, or works at or under the auspices of the University (including but not limited to students, adjunct and sessional faculty, librarians, program directors, post-doctoral fellows, emeriti and those holding a visiting appointment).
- 1.2 Please note, the requirement to file annual conflict of commitment and conflict of interest disclosures (as set out in Sections 3.3 and 4.4 of this Policy) only apply to those UBC Persons who are Faculty Members or who have signing authority over an active account that is used for a research purpose (see Section 1.1 of the Procedures to this Policy).

2. GeneralConflicts of Commitment

2.1 Identifying Conflicts of Commitment

- 2.1.1 <u>All UBC Persons are responsible for identifying, disclosing, and appropriately</u> managing any Conflict of Commitment in accordance with this Policy and conducting themselves with the highest ethical standards, including considering any Conflict of Commitment in a manner that will bear the closest scrutiny.
- 2.1.2 UBC has not attempted to develop an exhaustive list of Conflicts of Commitment as each situation is dependent upon its specific facts. However, the following are examples of situations where a Conflict of Commitment is deemed to exist:
 - (a) 2.1 A Conflict of Commitment occurs where a UBC Person engages in Non-University Activities that are substantial or demanding of the UBC Person's time and attention and adversely affect the discharge of the UBC Person's responsibilities to the University. UBC Persons must disclose Conflicts of Commitment and obtain written approval prior to engaging in any activity that may give rise to a Conflict of Commitment-where a UBC Person engages in a Non-University Activity during their normal work hours at UBC;
 - (b) where a UBC Person employed on a full-time basis by UBC enrolls in a degree or diploma program at UBC, or any other educational institution;
 - (c) where a UBC Person uses UBC resources for Non-University Activities;
 - (d) where a full-time Faculty Member's total involvement in Outside Professional Activities is greater than 52 days per year, including evenings, weekends and vacations; or
 - (e) where a full-time Faculty Member will be away from their place of work at UBC for a period of 30 consecutive days (excluding holidays) as a result of the Faculty Member's Outside Professional Activities.

2.2 Disclosing Conflicts of Commitment

- 2.2.1 If a UBC Person intends to engage in any Non-University Activities that could lead to a Conflict of Commitment, then the UBC Person must immediately disclose that activity to their Initial Reviewer pursuant to Section 3 of the Procedures. These disclosure obligations also apply to all UBC Persons as soon as knowledge of the Conflict of Commitment arises and continue until the Conflict of Commitment ceases to exist.
- 2.2.2 In addition, all Faculty Members and Staff Members are required to file annual Conflict of Commitment disclosures.

- 2.2.3 If a Faculty Member intends to engage in any activity set out in Section 2.1.2(d) or Section 2.1.2(e) of this Policy, then they must disclose and obtain prior written authorization from the Dean of the faculty in which they hold their primary appointment, a copy of which must be filed with the COI Manager.
- 2.2.4 All UBC Persons are encouraged to bring any questions or concerns regarding their Conflicts of Commitment to their Initial Reviewer, who may consult with the COI Manager.

2.3 Assessing and Managing Conflicts of Commitment

- 2.3.1 If an Initial Reviewer becomes aware of a Conflict of Commitment, whether through disclosure or otherwise, then they must assess the Conflict of Commitment taking into consideration the following:
 - (a) whether a UBC Person's engagement in the Non-University Activity interferes with their professional obligations to UBC, including the elements set forth in Section 1.3 of this Policy; and
 - (b) whether the benefit to UBC from a UBC Person's engagement in the Non-University Activity outweighs the interference with their professional obligations to UBC or would outweigh the interference if appropriately managed through a Management Plan.
- 2.3.2 Once the Initial Reviewer has assessed the Conflict of Commitment situation, the Initial Reviewer, may determine what action, if any, is appropriate pursuant to Section 4 of the Procedures.

3. 2.2 Conflicts of Interest and perceived

- 3.1 Identifying Conflicts of Interest can arise
 - 3.1.1 All UBC Persons are responsible for identifying, disclosing, and appropriately managing any Conflict of Interest in accordance with this Policy and conducting themselves at all times with the highest ethical standards, including considering any Conflict of Interest in a manner that will bear the closest scrutiny.
 - 3.1.2 Any UBC Person who engages in Non-University Activities is responsible for identifying any Conflict of Interests arising naturally from a UBC Person's their engagement inside and outside the in Non-University, and the mere existence of Activities.
 - 3.1.3 <u>Identifying</u> a Conflict of Interest or the perception of a Conflict of Interest does not necessarily imply wrongdoing on anyone's part. Nonetheless, Conflicts of Interest

and situations that give rise to perceptions of a Conflict of Interest must be recognized, disclosed, and assessed.involves two elements:

2.3

- (a) the UBC Person's or a Related Party's personal interest(s), including without limitation, business, commercial or Financial Interests and personal matters and career interests; and
- (b) the UBC Person's professional obligations to UBC.
- 3.1.4 A perceived Conflict of Interest is a situation where includes the following situations:
 - (a) an actual or potential Conflict of Interest may or may not exist, but where there may be, nonetheless, from the perspective of a reasonably well-informed, impartial observer, a perception of a Conflict of Interest. This is sometimes referred to as an "apparent" conflict of interest.
- 2.4 Conflicts of Interest and perceived Conflicts of Interest fall into one of two categories: those that are permissible if appropriately managed; and those that are prohibited because they cannot be appropriately managed.
 - i
 - (b) a situation where the UBC Person's personal interest has not yet conflicted but has the potential or might be expected to conflict with the UBC Person's professional obligations to UBC; or
 - (c) 2.5 UBC Persons are responsible for seeking guidance from their Initial Reviewers before engaging in any activity that may be questionable. If in doubt about whether a situation would, the perception of a Conflict of Interest from the perspective of a reasonably well-informed, impartial observer, appear to be a Conflict of Interest, the UBC Person must seek guidance from his or her Initial Reviewer.
 - 3.1.5 UBC has not attempted to develop an exhaustive list of Conflicts of Interest as each situation is dependent upon its specific facts. However, a non-exhaustive list of situations where a Conflict of Interest is deemed to exist is set forth in Section 2 of the Procedures.

3.2 Disclosing Conflicts of Interest

3.2.1 If a UBC Person intends to engage in an activity that could lead to a Conflict of Interest, then the UBC Person must immediately disclose that activity to their Initial Reviewer pursuant to Section 3 of the Procedures and to all parties affected by the Conflict of Interest. For example, if a UBC Person makes a public statement, in the media or otherwise, and identifies themselves as being associated with UBC, then the UBC Person must disclose, as part of that public statement, any other relevant <u>relationships, circumstances, or engagements in other Non-University Activities that</u> <u>give rise to a Conflict of Interest.</u>

2.6 Except as required by judicial process or law or Section 2.9 of this Policy, any information disclosed by a UBC Person to the University pursuant to this Policy will be held by the University in confidence and will only be available to those individuals who are responsible to the University for its review, for the management, administration and audit of the University, and to funding agencies where relevant to an application for research funding to that agency.

2.7-----

- 3.2.2 If prior disclosure is impossible, then disclosure must be made as soon as possible after commencement of the activity.
- 3.2.3 Prior to commencing a research project, a UBC Person who is a principal investigator or researcher must disclose all Conflicts of Interest that may arise in relation to the research project.
- 3.2.4 These disclosure obligations also apply to all UBC Persons as soon as knowledge of the Conflict of Interest arises and continue until the Conflict of Interest ceases to exist.
- 3.2.5 In addition, all Faculty Members and Staff Members are required to file annual Conflict of Interest disclosures.
- 3.2.6 All UBC Persons are encouraged to bring matters of concernary questions or concerns regarding their Conflicts of Interest and perceived Conflicts of Interest to the appropriate Administrative Head of Unit. In any case, any person who knows or suspects that a Conflict of Interest or a situation giving rise to a perception of a Conflict of Interest has gone unreported, or has not been managed properly, may contact the Chair of the COI Committee, in writing, care of the Conflict of Interest Administrator, whose contact information can be found on the Office of the University Counsel's website. The COI Committee will review the situation in an objective, timely fashion, respecting the rights of individuals in conformity with the principles of procedural fairness in the university context, and will provide its recommendations in a report to the Responsible Executive and to the Dean of the Faculty concerned or the Administrative Head of Unit concerned, as appropriate.
- 2.8 All UBC Persons are encouraged to bring matters of concern regarding Conflicts of Commitment to the appropriate Administrative Head of Unit. In any case, any person who knows or suspects that a Conflict of Commitment has gone unreported may contact the Conflict of Interest Administrator, in writing. Upon receipt, the Conflict of Interest Administrator will forward the complaint or concern to the Initial Reviewer or such other person as the Responsible Executive may deem appropriate, who will review the situation in an objective, timely fashion, respecting the rights of individuals in conformity with the principles of procedural fairness in the university context, and will notify the Responsible Executive and the Dean of the Faculty

concerned or the Administrative Head of Unit concerned, as appropriate, of the circumstances and the action(s) taken.to their Initial Reviewer, who may consult with the COI Manager.

3.3 Assessing and Managing Conflicts of Interest

- 3.3.1 If an Initial Reviewer becomes aware of a Conflict of Interest, whether through disclosure or otherwise, then they must assess the situation to determine what action, if any, is appropriate, including taking into consideration the following:
 - (a) the impact on the UBC Person's ability to satisfy their professional obligations to UBC, including the elements set forth in Section 1.3 of this Policy;
 - (b) <u>the degree to which the proposed action will be detrimental to the interests of</u> <u>UBC;</u>
 - (c) <u>the degree to which it may harm the reputation and integrity of the UBC</u> <u>Person(s) or UBC, or both;</u>
 - (d) <u>the degree to which it may compromise the professional judgment of any UBC</u> Person involved in conducting or reporting research;
 - (e) <u>the extent to which the proposed action or activity could be managed through</u> <u>an appropriate Management Plan; and</u>
 - (f) any of the other Conflict of Interest elements set forth in Section 2 of the Procedures.
- 3.3.2 Once the Initial Reviewer has assessed the Conflict of Interest situation, the Initial Reviewer, may determine what action, if any, is appropriate pursuant to Section 4 of the Procedures.

4. <u>Reporting of Complaints or Concerns</u>

- 4.1 All persons are encouraged to contact the COI Manager in writing if they believe that a Conflict of Interest or Conflict of Commitment has been unreported, misrepresented, unassessed, or unmanaged or that there has been any other breach of this Policy or any associated Procedures or Rules. Upon receipt, the COI Manager will handle the complaint or concern in accordance with Section 6.1.4 of the Procedures.
- 4.2 2.9-All complaints or concerns submitted pursuant to Section 2.7 and 2.8 of this Policy will be taken seriously. The anonymity of the person making the complaint or raising a concern will, unless otherwise directed by that person, be maintained, except as required by a legal process or the law. However, any information provided anonymously will not cannot be acted upon unless the information is independently substantiated. The University will protect personal information of all parties involved as required under the Freedom of Information and Protection of Privacy Act. The University will not tolerate anyAllegations of retaliation;

directly or indirectly, against anyoneany person who, in good faith, makes a complaint or raises a concern pursuant to Section 2.7 or 2.8 of this Policy, gives evidence, or otherwise participates in a process under this Policy will be handled in accordance with UBC's Retaliation Policy.

3. Disclosing Conflicts of Commitment

- 3.1 Where a UBC Person intends to engage in any activity that gives rise to a Conflict of Commitment, the UBC Person must disclose that activity and obtain the prior written approval of his or her Initial Reviewer.
- 3.2 The following are examples of situations where a Conflict of Commitment exists for Faculty Members, and despite the foregoing, may not be engaged in unless disclosed to and authorized by the Dean of the Faculty in which the Faculty Member has his or her primary appointment or such Dean's designate:
 - 3.2.1 Where a full-time Faculty Member's total involvement in Outside Professional Activities is greater than 52 days per year, including evenings, weekends and vacations.
 - 3.2.2 Where a full-time Faculty Member will be away from his or her place of University work for a period of 30 consecutive days (excluding holidays) as a result of the Faculty Member's Outside Professional Activities.
- 3.3 In addition, at least annually and whenever there is a material change from the previously submitted disclosure, any UBC Person identified in Section 1.1 of the Procedures to this Policy must disclose circumstances (if any) that give rise to a Conflict of Commitment, including, if applicable, the extent, nature, and timing of his or her Outside Professional Activities.

4. Disclosing Conflicts of Interest and Perceived Conflicts of Interest

- 4.1 In the University community, full and complete disclosure of Conflicts of Interest and perceived Conflicts of Interest is a key element in protecting the integrity of the University and UBC Persons and preserving the public's confidence in their ability to pursue and disseminate knowledge, devoid of bias and personal interests. Without that public confidence, the effectiveness of the University as a public institution and of UBC Persons as intellectual leaders is diminished. Therefore, UBC Persons must disclose Conflicts of Interest and perceived Conflicts of Interest to his or her Initial Reviewer and to all parties affected by the Conflict of Interest or perceived Conflict of Interest. For example, where a UBC Person makes a public statement and identifies himself or herself as being associated with the University, the UBC Person must disclose, as part of that statement, any other relevant relationships or circumstances that give rise to a Conflict of Interest or a perceived a Conflict of Interest.
- 4.2 Prior to commencing a research project, a UBC Person who is an investigator must disclose all Conflicts of Interest and perceived Conflicts of Interest that may arise in relation to the research project.

- 4.3 Prior to undertaking any activity that may give rise to a Conflict of Interest or a perceived Conflict of Interest, whether or not that activity is within the scope of the UBC Person's work at the University, a UBC Person must disclose that activity. Where prior disclosure is impossible, disclosure must be made as soon as possible after commencement of the activity.
- 4.4 In addition, at least annually and whenever there is a material change, any UBC Person identified in Section 1.1 of the Procedures to this Policy must disclose circumstances (if any) that give rise to a Conflict of Interest or a perceived Conflict of Interest.

5. Assessing and Managing Conflicts of Commitment

- 5.1 Where a UBC Person seeks approval for a Conflict of Commitment, the Initial Reviewer, Dean or Dean's delegate (as applicable) may, in his or her discretion, grant approval and may impose conditions upon such approval, taking into consideration the following:
 - 5.1.1 whether the activity interferes with the UBC Person's obligations to the University; and
 - 5.1.2 if so, whether the activity nonetheless contributes to or benefits the University in such a way that warrants the interference.
- 5.2 Where the Initial Reviewer, Dean or Dean's delegate (as applicable) determines that the activity is not permissible, the UBC Person must not proceed with the proposed activity.
- 5.3 Where the UBC Person does not agree with the decision regarding a Conflict of Commitment made by their Initial Reviewer, or in respect of a Conflict of Commitment described in Section 3.2 of this Policy, the Dean or the Dean's delegate (as applicable), the UBC Person may appeal the decision to the person holding the position immediately superior to the person who made the decision. The person who is to consider the appeal may, at his or her discretion, refuse to hear an appeal that is filed more than 30 days after the date of the Initial Reviewer's decision.
- 5.4 In addition, where the UBC Person has a joint appointment, the Initial Reviewer will be responsible to keep the other Head of Administrative Unit up to date with a copy of that UBC Person's disclosures and any applicable management protocol(s).
- 6. Role of Initial Reviewer in Assessing and Managing Conflicts of Interest and Perceived Conflicts of Interest
 - 6.1 When an Initial Reviewer becomes aware of a Conflict of Interest or a perceived Conflict of Interest, he or she will assess the situation to determine what action, if any, is appropriate.
 - 6.2 Where the Initial Reviewer determines that a proposed action or activity, including a proposed research project, is permissible as disclosed with no further action, the Initial Reviewer's determination will be recorded and the UBC Person may proceed with the action or activity, subject to obtaining all other applicable approvals.

- 6.3 Where the Initial Reviewer determines that an action or activity, including a proposed research project, is permissible only if appropriately managed, the Initial Reviewer will develop and implement a protocol to manage the Conflict of Interest or perceived Conflict of Interest (as the case may be) and the UBC Person, subject to obtaining all other applicable approvals, may proceed with the proposed activity provided that he or she complies with the management protocol.
- 6.4 Where the Initial Reviewer determines that an action or activity, including a proposed research project, is not permissible, the UBC Person must not proceed with the action or activity.
- 6.5 Where the UBC Person does not agree with a decision made by the Initial Reviewer pursuant to Sections 6.1 through 6.4, the UBC Person may appeal the decision in accordance with the internal review process (if any) established under Section 5 of the Procedures or to the COI Committee, if the unit has no such internal review process. The person or committee who is to consider the appeal may, at his, her or their discretion, refuse to hear an appeal that is filed more than 30 days after the date of the Initial Reviewer's decision.
- 6.6 Where the Initial Reviewer is part of the Conflict of Interest or the perceived Conflict of Interest, or otherwise if the Initial Reviewer deems it appropriate or necessary, the Initial Reviewer will refer the disclosure to either:
 - 6.6.1 if the UBC Person concerned is a Faculty Member, his or her Dean or the Dean's delegate; or
 - 6.6.2 if the UBC person is not a Faculty Member, such other person as may be determined by
- 4.3 UBC will protect the personal information of all parties involved as required under applicable legislation, including the *Freedom of Information and Protection of Privacy Act* (British Columbia). Except as required by a legal process or law, any personal information disclosed by a UBC Person pursuant to this Policy will be held in confidence and will only be available to relevant granting or funding agencies and those with a need to know the information in order to fulfill their responsibilities to UBC, including:
 - 4.3.1 those responsible under this Policy and any associated Procedures or Rules for the review, management, or audit of Conflict of Interest and Conflict of Commitment disclosures; or
 - 4.3.2 <u>those who are involved in investigations of alleged breaches of UBC Policies or other</u> <u>misconduct, which may include: Central Human Resources; the Office of the Vice</u> <u>President, Research and Innovation; the Office of the University Counsel,</u>
 - and that person shall take on the role of Initial Reviewer. If the UBC Person concerned is a Faculty Member and the Faculty has an internal review process, then the Dean or the Dean's delegate shall act in accordance with such process. If it is not appropriate

for either the Dean or the Dean's delegate to take on the role of Initial Reviewer, the matter will be referred to; members of the COI Committee; the Responsible Executive for a determination of who should take on the role of Initial Reviewer.

7. Role of the COI Committee in Assessing and Managing Conflicts of Interest and Perceived Conflicts of Interest

- 7.1—The COI Committee is charged with the following responsibilities:
 - 7.1.1 to assess disclosures of Conflicts of Interest and perceived Conflicts of Interest to determine whether, subject to obtaining all other applicable approvals:
 - (a) the proposed action or activity, including a proposed research project, is permissible as disclosed with no further action;
 - (b) the proposed action or activity, including a proposed research project, is permissible only if a protocol is implemented to ensure that the Conflict of Interest or perceived Conflict of Interest (as the case may be) is appropriately managed; or
 - (c) the proposed action or activity, including a proposed research project, is not permissible;
 - 7.1.2 to serve as the final avenue of appeal (such as where a UBC Person disagrees with the assessment of an Initial Reviewer or the assessment resulting from an internal process established by a Dean under Section 5 of the Procedures to this Policy);
 - 7.1.3 to audit random samples of disclosures reviewed and determined to be permissible by Initial Reviewers and by those reviewing disclosures in accordance with the process established within a particular unit. The COI Committee will also periodically audit random samples of management protocols implemented in accordance with Section 7.3 below;
 - 7.1.4 to educate UBC Persons on Conflicts of Interest, perceived Conflicts of Interest, Conflicts of Commitment and the application of this Policy;
 - 7.1.5 to review concerns, whether self-initiated or otherwise, that a Conflict of Interest or a situation giving rise to a perception of a Conflict of Interest has gone unreported, or has not been managed properly; and
 - 7.1.6 to fulfill such other roles and responsibilities assigned to it by any other policy of the University, or by the University Counsel.
- 7.2 Where the COI Committee determines that a proposed action or activity, including a proposed research project, is permissible as disclosed with no further action, the UBC Person may proceed with the proposed action or activity.

- 7.3 Where the COI Committee determines that a proposed action or activity, including a proposed research project, is permissible only if a management protocol is implemented to ensure that the Conflict of Interest or perceived Conflict of Interest (as the case may be) is appropriately managed, the UBC Person may only proceed with the proposed action or activity in accordance with the management protocol.
 - 7.4 Where the COI Committee determines that an action or activity, including a proposed research project, is not permissible, the UBC Person must not proceed with the proposed action or activity. ; or any of their respective and applicable delegates.



PROCEDURES ASSOCIATED WITH THE POLICY

Pursuant to the Regulatory Framework Policy, the President may approve Procedures or the amendment or repeal of Procedures. Such approvals must be reported at the next meeting of the UBC Board of Governors or as soon thereafter as practicable.

<u>Capitalized terms used in the Policy that are not otherwise defined herein shall have the meanings given</u> to such terms in these Procedures or Rules.

1. 8. Definitions

- 1.1 8.1 "Administrative Head(s) of Unit" means a Director of a service an administrative unit; a Head of an academic department; a Director of a centre, institute or school; a Principal of a college; a Dean, an Associate Vice-President, of a faculty; the Registrar; the University Librarian; a Vice-President; or the President or the equivalent; or any of their respective delegates, which may include Associate Vice-Presidents; Vice-Principals; and Associate Vice-Principals.
- 1.2 8.2 "Business" means a corporation, partnership, sole proprietorship, firm, franchise, association, organization, holding company, joint stock company, receivership, business or real estate trust, or other legal entity organized for profit, not for profit, or charitable purposes, but excluding the University, an affiliated Hospital, a private medical practice, or other entity controlled by, controlling, or under common control with the University or an affiliated Hospital.
- 1.3 <u>"Central Human Resources</u>" means Employee and Labour Relations or Faculty Relations on the UBC Vancouver campus, or Human Resources on the UBC Okanagan campus.
- 1.4 8.3–"COI Committee" means a committee established by the University CounselResponsible Executive to carry out the roles and responsibilities set out in Section 7 of thisthe Policy, and which includes at least one independent person who is neithernot a Faculty Member nor a staff member of the UniversityUBC Person.

8.4

1.5 <u>"COI Manager" means the individual appointed by the Responsible Executive to oversee the</u> administration and governance of the Policy or other duties as delegated by the Responsible Executive from time to time, and, whose contact information can be found on the website of the office of the Responsible Executive.

- 1.6 **"Conflict(s) of Commitment**" has the meaning given to it in Section 2.1 of this Policymeans a situation where a UBC Person engages in Non-University Activities that adversely affect the discharge of the UBC Person's primary responsibilities to UBC.
- 1.7 8.5-"Conflict(s) of Interest" means a situation in which a UBC Person, or his or hertheir Related Party(ies), has a personal interest that <u>actually</u> conflicts or, could conflict with the UBC Person's obligations to the University. Please note:
 - 8.5.1 The existence of a Conflict of Interest involves two elements: the UBC Person's or a Related Party's personal interest(s), and the UBC Person's obligations to the University. Personal interests may include business, commercial or financial interests, as well as personal matters and career interests. A UBC Person's obligation to the University is to act in the University's interests, which includes acting in support of the University's integrity and fundamental mission and avoiding circumstances that may undermine public confidence and trust. The University has many diverse objectives, which include many supporting commitments, goals and actions. From time to time these objectives, commitments, goals and actions may appear to be, or may actually be in conflict with one another. In such cases, the University's overriding interest is that its activities are undertaken with integrity and in accordance with the highest ethical standards.

8.5.2 Conflicts of Interest may be actual or potential.

- (a) An actual Conflict of Interest is a situation where the personal interest actually, or there is a perception that it conflicts with the UBC Person's obligations to the University.
- (b) A potential Conflict of Interest is a situation where the personal interest has not yet conflicted with the UBC Person's obligations to the University, but might be expected to <u>UBC</u>.
- 8.6 "Executive Position" means any position that includes responsibilities for a material function of the operation or management of a Business.
- 1.8 8.7 "Faculty Member" means, for the purposes of this Policy, a person employed an individual appointed by the University as ain the following faculty titles, ranks and descriptions: professor, associate professor, assistant professor, professor of teaching, assistant professor of teaching, associate professor of teaching, adjunct professor, clinical professor, clinical associate professor, clinical assistant professor, clinical instructor, instructor II, senior instructor, twelve monthclinical fellow, academic program director, librarian, research associate, post-doctoral fellow, sessional lecturer, lecturer, visiting assistant professor, visiting associate professor, visiting professor, visiting lecturer, visiting scientist, visiting scholar, emeritus, honorary faculty or affiliate appointment or in an equivalent position designated by the applicable senate.

1.9 8.8 "Financial Interest" means having or having the expectation to receive:

- 1.9.1 8.8.1 greater than a 5% ownership interest in a single entity Business; or
- **1.9.2 8.8.2** anything with a monetary value exceeding \$5,000 in any one calendar year, including remuneration (e.g., salary, consulting fees, retainers, honoraria, bonuses, gifts, speaker's fees, advisory board remuneration, finders or recruitment fees), equity interests (e.g., stocks, stock options or other ownership interests), and intellectual property rights (e.g., patents, copyrights, royalties or other payments from such rights).
- 1.10 "Guidelines" has the meaning set forth in UBC's Regulatory Framework Policy.
- **1.11 8.9—"Initial Reviewer(s)"** means the individual responsible for <u>initially</u>-reviewing a UBC Person's disclosures of Conflicts of Interest <u>and perceivedor</u> Conflicts of <u>Interest or approving</u> a UBC Person's Conflicts of Commitment. Except as otherwise designated by the University Counsel, a UBC Person'sFor Faculty Members, the Initial Reviewer is <u>normally</u> the Administrative HeadDean of Unitthe faculty in which that UBC person holds his or herthey hold their primary appointment—, unless their faculty is departmentalized, in which case it is normally the Head of their department or school. For Staff Members, the Initial Reviewer is <u>normally</u> their immediate supervisor and, where the Staff Member has more than one immediate supervisor, the Initial Reviewer is the immediate supervisor designated by the Responsible Executive. For all other UBC Persons, the Initial Reviewer is normally their immediate supervisor. Notwithstanding the foregoing, the Responsible Executive may designate a different Initial Reviewer for any UBC Persons.
- 1.12 "Management Plan" means the written document governing and managing a UBC Person's participation in an activity deemed to constitute a Conflict of Interest or Conflict of Commitment.
- 1.13 8.10-"Non-University Activity(ies)" means any activity outside a UBC Person's scope of work with the UniversityUBC and includes Outside Professional Activities.
- 1.14 8.11 "Outside Professional Activity(ies)" means any activity outside a UBC Person's scope of work with the UniversityUBC that involves the same specialized skill and knowledge that the UBC Person utilizes in his or hertheir work with the UniversityUBC and includes the operation of a Business, consulting or advisory services, external teaching, external academic appointments and external speaking engagements.
- 1.15 <u>"Policy" means the COI Policy.</u>
- **1.16 <u>"Position of Authority" means any position that includes responsibilities for a material</u> function of the operation or management of a Business.**
- **1.17 8.12** "**Related Party**(<u>ies</u>)" means a UBC Person's immediate family member (e.g., partner, <u>or sibling</u>), <u>direct descendant or ancestor (e.g., child, grandchild, parent</u>, or <u>siblinggrandparent</u>), <u>or any</u> other person living in the same household, any other person with

whom the UBC Person shares a Financial Interest, either directly or indirectly, or any entity<u>Business</u> in which the UBC Person has an ownership interest of more than 5%.

8.13 "UBC Person" means all individuals identified in Section 1.1 of this Policy.

T



PROCEDURES ASSOCIATED WITH THE COLPOLICY

Pursuant to the Regulatory Framework Policy, the President may approve Procedures or the amendment or repeal of Procedures. Such approvals must be reported at the next meeting of the UBC Board of Governors or as soon thereafter as practicable.

Capitalized terms used in these Procedures that are not otherwise defined herein shall have the meanings given to such terms in the accompanying Policy, being the COI Policy.

1. Annual Disclosures

- 1.18 "Related Policy(ies)" means the related policies specific to the Policy as identified in the official repository for the Policy maintained by the Office of the University Counsel and published on its website.
- 1.19 <u>**"Respondent"**</u> means the UBC Person who is the subject of an investigation of an alleged breach of the COI Policy or associated Procedures or Rules.
- 1.20 **"Responsible Executive"** means individual(s) assigned by the President, pursuant to UBC's Regulatory Framework Policy, to be responsible for the Policy; and any sub-delegate of that assigned responsible individual(s) except to the extent that the power to delegate is specifically excluded in the Policy or in the appointment by the President.
- 1.21 "Rules" has the meaning set forth in UBC's Regulatory Framework Policy.
- 1.22 "Staff Member" means individual(s) employed by UBC in any of the following employee categories or titles:
 - 1.22.1 Management and Professional;
 - 1.22.2 Director of a service unit;
 - 1.22.3 Administrative AVP (as defined in UBC's Administrative AVPs Policy);
 - 1.22.4 Administrative VP (as defined in UBC's Administrative VPs Policy); and
 - 1.22.5 1.1 At least annually, any Any other UBC Person(s), who is not a Faculty Member-or, as required by the Responsible Executive (e.g., those who has have signing authority over an active account that is used for a research purpose, will submit a completed disclosure to his or her Initial Reviewer that discloses the following:UBC financial accounts).

1.1.1 all Outside Professional Activities;

1.1.2 all circumstances (if any) that give rise to a Conflict of Commitment;

- 1.23 "Tri-Agency" means the Canadian Institutes of Health Research, the Natural Sciences and Engineering Research Council, and the Social Sciences and Humanities Research Council, collectively.
- 1.24 **"UBC Person(s)**" means all Faculty Members, all Staff Members, all other employees of UBC, all students to the extent that they are carrying out UBC research, and all individual(s) who have agreed to carry out duties on behalf of UBC as a volunteer.
- 2. <u>1.1.3 all circumstances (if any) that give rise to a</u><u>Additional</u> Conflict of Interest or a perceived Conflict of Interest; and <u>Considerations:</u>
 - **1.1.4** such additional information asked for in the prescribed annual disclosure form, as it may be amended from time to time.
 - 1.2 Annual disclosures are to be made utilizing the Conflict of Interest / Conflict of Commitment module of the Researcher Information Services (RISe) system.

2. Recognizing Conflicts of Commitment

- 2.1 The following are examples of situations where a Conflict of Commitment may exist.
 - 2.1.1 Where a UBC Person engages in Non-University Activities during his or her normal work hours at the University.
 - 2.1.2 Where a UBC Person employed on a full-time basis by the University enrols in a degree or diploma program at the University, or any other educational institution.
 - 2.1.3 Where a UBC Person uses University resources for Non-University Activities.

3. Recognizing Conflicts of Interest and Perceived Conflicts of Interest

- 3.1 As each situation depends upon its specific facts, the University has not attempted to develop an exhaustive list of Conflicts of Interest. Rather, UBC Persons must conduct themselves at all times with the highest ethical standards in a manner that will bear the closest scrutiny. The following are examples a non-exhaustive list exists deemed to exist:
 - 2.1.1 <u>3.1.1 Wherewhere</u> a UBC Person's responsibility to instruct—and, evaluate and supervise_students in a fair, unbiased and effective manner is or could be impeded

or compromised. The <u>(Note, the</u> inherent power imbalance that exists between a UBC Person and a student must not be used for personal benefit.);

3.1.2 Where

2.1.2 where a UBC Person or a Related Party of the UBC Person-has a Financial Interest in the outcome of his or her research. In the area of their research, vigilance is required. The University's commitment to liaise with industry and to transfer technology, the growth of industry sponsored research, and the substantial increase in the University's interactions with outside organizations generally all lead to a corresponding increase in the potential for Conflicts of Interest. In addition, various governmental and other agencies that fund research are imposing increasingly stringent Conflicts of Interest requirements on investigators and institutions applying for research funding. UBC Persons must comply with this Policy and any applicable Conflicts of Interest policies imposed by other organizations, agencies or institutions. including any Financial Interest in a company sponsoring research at their UBC laboratory or

facilities;

- 2.1.3 3.1.3 Wherewhere a UBC Person or a Related Party has a Financial Interest in their teaching activities at the UniversityUBC, other than their annual salary from the University.UBC;
- 2.1.4 3.1.4 Wherewhere a UBC Person has influence over a decision about a proposed relationship between the University UBC and a Business in which the UBC Person or his or hera Related Party has a Financial Interest or holds an Executive Position. of Authority;
- 2.1.5 3.1.5 Wherewhere a UBC Person or his or hera Related Party has or obtains a Financial Interest or an Executive Position of Authority in a Business with which the UniversityUBC has an existing relationship and the Business is related to the UBC Person's work at the University.UBC;
- 2.1.6 where a UBC Person transfers research results or intellectual property arising from their UBC research to a Business in which the UBC Person or a Related Party has a Financial Interest or holds a Position of Authority;
- 2.1.7 3.1.6 Wherewhere a UBC Person is in a position to influence human resource decisions (such as recruitment, offer of employment, <u>appointment or selection process</u>, evaluation of performance, promotion, granting of tenure, <u>discipline</u>, <u>investigations</u>, or termination of employment) or admission decisions with respect to a person who is a Related Party or who has another relationship with whom the UBC Person has a relationship that might reasonably be perceived as creating a Conflict of Interest-;
- 2.1.8 3.1.7 Where where a UBC Person uses his or herattempts to use their position with the University UBC to solicit students, other UBC Persons, government agencies, private companies, or members of the public for Non-University Activities, including

employing or soliciting employment from students and or other UBC Persons forto perform services in a for the UBC Person or a Related Party on personal or commercial matter.matters;

- 2.1.9 3.1.8 Where where a UBC Person uses information that is acquired as a result of his or her relationshiptheir position with the UniversityUBC and not in the public domain for Non-University Activities, unless the UBC Person has proprietary rights (usually enforceable through copyright) to that information. (Note, UBC Persons should also be aware that insider trading restrictions may also apply to them. under the laws of British Columbia or other jurisdictions.);
- 2.1.10 3.1.9 Wherewhere a UBC Person's obligations to a board of directors, advisory boards, or the like of an outside organization interfere with or compromise the UBC Person's obligations to the University.UBC;
- 2.1.11 3.1.10 Wherewhere a UBC Person accepts gifts with a value of \$500 or more that would not have been offered if not for the UBC Person's position at the University, or work within the scope of a UBC Person's employment with the University.or role at UBC;
- 2.1.12 3.1.11 Where where a Faculty Member enrolsenrolls as a student in his or hertheir own Faculty.faculty; or
- 2.1.13 3.1.12 Where where personal considerations <u>could be reasonably perceived to</u> compromise a Faculty MemberUBC Person's professional judgment in conducting or reporting research, teaching, or carrying out administrative activities at UBC.

3. Required Disclosures

- 3.1 All UBC Persons must:
 - 3.1.1 immediately disclose to their Initial Reviewer if a UBC Person intends to engage in any Non-University Activities that could lead to a Conflict of Interest and Conflict of Commitment;
 - 3.1.2 4. Assessing and Managing immediately disclose Conflicts of Interest, Perceived and Conflicts of Commitment to their Initial Reviewer as soon as such situation or event arises; and
 - 3.1.3 <u>continue to disclose and submit such information, as requested, or at least annually</u> <u>until the Conflicts of Interest and Conflicts of Commitment cease to exist.</u>

4.1 Annual

3.2 <u>At least annually, all Faculty Members and Staff Members must submit a completed disclosure</u> that discloses the following:

- 3.2.1 <u>all Non-University Activities;</u>
- 3.2.2 <u>all circumstances (if any) that give rise to a Conflict of Interest or Conflict of</u> <u>Commitment; and</u>
- 3.2.3 <u>such additional information asked for in the prescribed annual disclosure form, as it</u> may be amended from time to time.
- 3.3 <u>All</u> disclosures will be subject to a preliminary evaluation conducted automatically against the pre-set parameters made by Faculty Members under Section 3.2 of these Procedures must be submitted utilizing the Conflict of Interest or the Conflict of Commitment module in the *RISe* system, which may be revised from time to time.
- 4.2 Without limiting the discretion of a person assessing a disclosure to consider all relevant factors, the following factors must be considered in assessing all disclosures:
 - 4.2.1 the impact on the UBC Person's ability to satisfy his or her obligations to the University;
 - 4.2.2 the degree to which the proposed action will be detrimental to the interests of the University, or in the research context, the degree to which it may compromise an investigator's professional judgment in conducting or reporting research; and
 - 4.2.3 the extent to which the proposed action or activity may be managed through an appropriate management protocol.
- ÷
- 3.4 All disclosures made by Staff Members under Section 3.2 of these Procedures must be submitted utilizing the Conflict of Interest or the Conflict of Commitment module in the Workday system.
- 3.5 All disclosures made by UBC Persons under Section 3.1 of these Procedures who are not Faculty Members or Staff Members must be submitted utilizing the Conflict of Interest or Conflict of Commitment module in the system deemed most appropriate by the Initial Reviewer.
- 4. Role of Initial Reviewer
 - 4.1 <u>The Initial Reviewer is responsible for assessing the disclosure and determining whether there</u> <u>are any Conflicts of Interest or Conflicts of Commitment, and, if so, whether they can and</u> <u>should be permitted subject to a Management Plan.</u>
 - 4.2 4.3 Where a person who reviews a disclosure is unable to determine If the Initial Reviewer determines that a proposed action or activity, including a proposed research project, is :

- 4.2.1 permissible <u>as disclosed</u> with no further action, <u>then</u> the <u>disclosure must be referred</u> to the next level of review.
- 4.4 Where a disclosure in relation to a Conflict of Interest or a perceived Conflict of Interest is referred to the COI Committee, the COI Committee will complete a committee report for each disclosureInitial Reviewer's determination will be recorded in writing and provide the UBC Person may proceed with a copy of the committee report.
- 4.5 Where the COI Committee determines that a management protocol is appropriate in relation to a Conflict of Interest or a perceived Conflict of Interest, the Initial Reviewer will monitor compliance with the management protocol.
- 4.6 Where the COI Committee determines that a Conflict of Interest exists, the COI Committee will:
 - 4.6.1 provide the Responsible Executive with any additional information shared by the UBC Person that is not part of the disclosure;
 - 4.6.2 disclose the existence of the Conflict of Interest to a funding agency, where relevant to an application for research funding to that agency; and
 - 4.6.3 disclose the existence of the Conflict of Interest to the applicable research ethics board where the Conflict of Interest relates to a research project involving human participants.
 - 4.7 Where the COI Committee determines that the proposed action or activity is <u>subject</u> to obtaining all other applicable approvals;
 - 4.2.2 permissible so long as the Conflict of Interest or a perceived Conflict of Interest isonly if appropriately managed, the COI Committee may itself create, amend and/or oversee through the implementation of the management protocol, or may instructa Management Plan, then the Initial Reviewer to do so.will develop and implement a Management Plan to manage the Conflict of Interest or the Conflict of Commitment and the UBC Person may proceed with the action or activity pursuant to the Management Plan; or

4.8 The forms of all disclosures and reports described in these Procedures will be prescribed by the Office of the University Counsel in consultation with the academic community.

- 4.2.3 <u>not permissible, then the UBC Person must not proceed with the proposed action or</u> activity giving rise to the Conflict of Interest or Conflict of Commitment.
- 4.3 <u>The Initial Reviewer may consult with the COI Manager in relation to any decisions required to</u> <u>be made under Section 4 of these Procedures, including the development of any applicable</u> Management Plan(s) and the COI Manager may in turn consult with the COI Committee.

- 4.4 If the Initial Reviewer is unable to determine whether a Conflict of Interest or a Conflict of Commitment or any proposed action or activity is permissible due to their own Conflict of Interest or otherwise, then the Initial Reviewer must refer the disclosure one level up for handling. If such a referral is not appropriate, then the matter will be referred to the Responsible Executive for a final determination of who should handle the matter.
- 4.5 Where the UBC Person is a Faculty Member who holds an appointment in more than one faculty, the Initial Reviewer must keep the Dean of the other faculty up-to-date with a copy of that Faculty Member's disclosures and any applicable Management Plan(s). Where the UBC Person is a Staff Member who has more than one immediate supervisor, then the Initial Reviewer must consult with and keep the other immediate supervisors up-to-date with a copy of that Staff Member's disclosures and any applicable Management Plan(s).

5. InternalUnit-Specific Review Process for Conflicts of Interest and Perceived Conflicts of Interest

- 5.1 A Dean, in consultation with the Office of the University Counsel, may establish an internal review process within his or her Faculty for the assessment of Conflicts of Interest and perceived Conflicts of Interest as appropriate for his or her Faculty. Generally, a level of review within each Faculty is desirable to facilitate the expeditious assessment of matters and ensure awareness of and vigilance to Conflict of Interest and perceived Conflict of Interest issues by all UBC Persons within that Faculty. For example, the appropriate process may commence with a disclosure being reviewed by the Administrative Head of Unit (as Initial Reviewer), then being referred to the Dean, and finally being referred to the COI Committee, as necessary. A Dean, in consultation with the Office of the University Counsel, may establish additional reporting requirements for Faculty Members in his or her Faculty. Provision may be made for those cases where a UBC Person disagrees with the assessment of the Initial Reviewer.
- 5.2 In establishing an appropriate internal review process, the Dean should consider the following factors:

5.2.1 the volume of disclosures the Faculty generates; and

5.2.2 the need for specialized knowledge and expertise in the academic/professional discipline to assess Conflicts of Interest or perceived Conflicts of Interest.

6. Guidelines for Conflicts of Commitment

- 6.1 Where an Initial Reviewer is the Head of an academic department, the Initial Reviewer may issue guidelines that address Conflicts of Commitment in his or her academic department (other than those described in Section 3.2 of the Policy), provided that such guidelines are in accordance with this Policy and Procedures. The guidelines may:
 - 6.1.1 provide for streamlined disclosure and approval procedures; and
 - 6.1.2 establish conditional pre-approvals for defined activities and categories of activities that may give rise to Conflicts of Commitment.

- 6.2 Guidelines may only be issued with the prior consent of the Dean of the Initial Reviewer's Faculty.
- 6.3 Guidelines are only in effect for so long as the Initial Reviewer who issued the guidelines acts as Initial Reviewer—for greater certainty, guidelines adopted by an Initial Reviewer are no longer valid when they leave that position, and previous guidelines do not bind the succeeding Initial Reviewer.
- 6.4 The Initial Reviewer must keep an up to date version of the guidelines filed with the Conflict of Interest Administrator at all times.

7. Advisory Role of Office of the University Counsel

5.1 <u>Notwithstanding Section 4 of these Procedures, if the UBC Person disclosing a Conflict of</u> Interest or a Conflict of Commitment is part of an academic or administrative unit for which unit-specific review processes have been established pursuant to one or more Rules, then the review shall be undertaken in accordance with such Rules.

6. Role of the COI Manager

- 6.1 The COI Manager is charged with the following responsibilities:
 - 6.1.1 **7.1** Anserving as the secretary to the COI Committee;
 - 6.1.2 <u>providing advice and guidance to</u> Initial <u>ReviewerReviewers</u>, Administrative <u>HeadHeads</u> of Unit-or the COI Committee may at any time seek the advice of the <u>Office of the University Counsel</u>, the Responsible Executive, and the COI Committee with respect to any matter pertaining to <u>thisthe</u> Policy and any associated <u>Procedures and Rules</u>, including without limitation, :
 - (a) the establishment of a process processes for the assessment of Conflicts of Interest and perceived Conflicts of Interest, Commitment;
 - (b) the assessment of any disclosure disclosures of Conflicts of Interest and Conflicts of Commitment; and
 - (c) the development of appropriate protocols for managing Conflicts of Interest and perceived Management Plan(s);
 - 6.1.3 making recommendations to the Responsible Executive in relation to the issuance of any Rules, Guidelines, templates, or other materials or resources and consulting the COI Committee as appropriate;

- 6.1.4 <u>handling any complaints or concerns received pursuant to Section 4.1 of the Policy</u> <u>as the COI Manager deems appropriate, including forwarding such complaints or</u> <u>concerns:</u>
 - (a) to the appropriate Initial Reviewer or any other person identified in a Rule establishing a unit-specific review process contemplated by Section 5 of these Procedures for assessment and management pursuant to either Section 2.3 or Section 3.3 the Policy; or
 - (b) <u>to the attention of the COI Committee for handling pursuant to Section 9 of</u> <u>these Procedures; and</u>
- 6.1.5 <u>providing training to UBC Persons regarding Conflicts of Interest, Conflicts of</u> <u>Commitment, and the application of the Policy and any associated Procedures and</u> <u>Rules.</u>

7. Role of the COI Committee

- 7.1 The Responsible Executive must establish a COI Committee.
- 7.2 The COI Committee is charged with the following responsibilities:
 - 7.2.1 <u>auditing random samples of</u> Conflicts of Interest and Conflicts of Commitment that have been approved and any related Management Plans for quality control purposes and advising Initial Reviewers on appropriate standards for the quality of disclosures, reviews, and Management Plans;
 - 7.2.2 <u>acting as a sounding board and source of advice regarding the appropriate handling</u> of specific disclosures or other cases referred to it by the COI Manager.
 - 7.2.3 <u>serving as the final avenue of appeal (such as where a UBC Person disagrees with the assessment of an Initial Reviewer or the assessment resulting from a Rule established under Section 11 of these Procedures);</u>
 - 7.2.4 to review concerns or issues in accordance with Section 9 of these Procedures, whether self-initiated or brought forth by the COI Manager pursuant to Section 6.1.4(b) of these Procedures; and
 - 7.2.5 <u>fulfilling such other roles and responsibilities as may be assigned to it by the</u> <u>Responsible Executive.</u>

8. SanctionsAppeals

8.1 This Policy is meant to protect both the University and UBC Persons. The Administrative Head of Unit will notify the University Counsel and take immediate and appropriate action when he or she becomes aware of violations of this Policy or the Procedures.All UBC Persons may appeal any decision regarding a Conflict of Interest or Conflict of Commitment made by their Initial Reviewer in accordance with Section 4 of these Procedures by sending a notice of appeal to the COI Manager in writing using such form as may be prescribed by the COI Manager. The COI Manager will bring the notice to the attention of the COI Committee.

- 8.2 <u>The COI Committee may, at its discretion, refuse to hear an appeal that is filed more than 30</u> days after the date of the Initial Reviewer's decision.
- 8.3 The Chair of the COI Committee has the discretion to determine the manner in which the COI <u>Committee considers appeals, provided that such discretion must not be exercised in a</u> <u>manner that is inconsistent with the Policy or any associated Procedures or Rules or any</u> <u>directions issued by the Responsible Executive.</u>

9. Referral of Complaints or Concerns to the COI Committee

- 9.1 If a complaint or concern is forwarded to the attention of the COI Committee by the COI Manager pursuant to Section 6.1.4(b) of these Procedures, the COI Committee will normally make an initial determination as to whether the alleged breach has resulted from an honest and reasonable error or oversight. The COI Committee may reconsider its initial determination at any time.
- 9.2 Where the COI Committee's initial determination is that the alleged breach has resulted from an honest and reasonable error or oversight, the COI Committee will normally advise or direct the COI Manager, the Respondent, their Initial Reviewer and any other person that the COI Committee determines should receive such advice or directions as to the appropriate handling of the matter, and such persons must comply with any such directions.
- 9.3 If the COI Committee makes an initial determination that the alleged breach is not the result of an honest and reasonable error or oversight, then the COI Committee will normally investigate it pursuant to Section 9.4 of these Procedures.
- 9.4 The Chair of the COI Committee has the discretion to determine whether and how the COI Committee carries out its investigations, provided that such discretion must not be exercised in a manner that is inconsistent with the Policy or any associated Procedures or Rules or any directions issued by the Responsible Executive. Without limiting the generality of the foregoing, the Chair of the COI Committee may:
 - 9.4.1 determine that the investigation be terminated;
 - 9.4.2 recommend to the Responsible Executive that the investigation be undertaken as a Consolidated Investigation, as defined in UBC's Investigations Policy; and
 - 9.4.3 request that the Responsible Executive appoint an external investigator or other person to undertake the investigation on behalf of the COI Committee.
- 9.5 Unless otherwise determined by the Chair of the COI Committee, investigations of complaints or concerns by the COI Committee will include:

- 9.5.1 <u>consulting with Central Human Resources, UBC administrators, and others, as</u> <u>appropriate;</u>
- 9.5.2 requesting written submissions or interviewing the Respondent and other parties with information that may be relevant to the alleged breaches, including the party who submitted the complaint or concern; and
- 9.5.3 <u>obtaining documents relevant to the alleged breach.</u>
- 9.6 Upon completion of an investigation of the alleged breach, the COI Committee will prepare and deliver to the Responsible Executive a written report that includes the following information:
 - 9.6.1 <u>a description of the alleged breach;</u>
 - 9.6.2 <u>the names of the parties who provided information and a summary of the</u> information they provided;
 - 9.6.3 <u>a summary of the relevant documents and other material reviewed;</u>
 - 9.6.4 <u>findings of fact based on the information, documents, and other material gathered</u> <u>during the investigation;</u>
 - 9.6.5 <u>a determination as to whether a breach of the Policy or any associated Procedures</u> or Rules have occurred; and
 - 9.6.6 <u>if a breach is determined to have occurred, then any mitigating or aggravating</u> <u>circumstances the COI Committee considers to be relevant.</u>

10. Outcomes Resulting from Breaches

- 10.1 8.2 Sanctions If, pursuant to Section 9.6 of these Procedures, the Responsible Executive receives a report from the COI Committee setting out a determination by the COI Committee that a breach of the Policy or any associated Procedures or Rules has occurred, the Responsible Executive will immediately forward the report to the Dean of the Respondent's faculty if the Respondent is a Faculty Member and to the Respondent's immediate supervisor if the Respondent is a Staff Member, with a copy to Central Human Resources. The Respondent's Dean or immediate supervisor will be commensurate then consult with the seriousness of the violation. AnyCentral Human Resources to receive advice on what, if any, disciplinary sanctionaction is appropriate. Any disciplinary action that is imposed on a UBC Person pursuant to this Policy or another UBC policy will be in accordance with the applicable terms and conditions of employment, including any applicable collective agreement, or such other contractual process(es) applicable to the UBC Person.
- 10.2 8.3-In addition to any disciplinary sanctions action, where a researcher Faculty Member fails to comply with this the Policy or these Procedures, the Dean or the Initial Reviewer or the relevant (if the Initial Reviewer is not the Dean) may direct Financialeither the Office of

<u>Research</u> Services or the University-Industry Liaison Office to freeze that researcher's the research account(s), of the Faculty Member until further notice and the University. UBC may also terminate access to funding or require the repayment of any financial benefit gained by the researcher Faculty Member.

10.3 Failure to comply with the Policy and any associated Procedures and Rules could also lead to reporting by the Office of the Vice-President, Research and Innovation to the applicable Tri-Agency, or the U.S. Department of Health and Human Services, or any other funding and granting agencies and organizations and potentially result in the loss of research funding from such funding and granting agencies and organizations.

11. Establishing Guidelines and Rules

- 11.1 The Responsible Executive has the authority to establish, amend, and repeal Rules that address Conflicts of Commitment and Conflicts of Interest, provided that such Rules are not inconsistent with the Policy or any associated Procedures. The Responsible Executive must provide the Office of the University Counsel with a notice of the establishment, amendment, or repeal of any such Rules so that they can be reflected in the official repository of Board Policies maintained by the Office of the University Counsel pursuant to UBC's Regulatory Framework Policy. The establishment, amendment, or repeal of any Rules take effect and are binding on UBC Persons when approved by the Responsible Executive or on such later date as may be specified by the Responsible Executive.
- 11.2 The Responsible Executive may issue Guidelines as contemplated by UBC's Regulatory Framework Policy. Such Guidelines may address Conflict of Commitment or Conflict of Interest issues of interest to UBC Persons across all of UBC or to various subsets of UBC. Where an Administrative Head of Unit considers that Guidelines specific to their unit would be beneficial, the Administrative Head of Unit may recommend that the Responsible Executive issue such unit-specific Guidelines. The Responsible Executive must provide the Office of the University Counsel with a notice of the establishment, amendment, or repeal of any Guidelines so that they can be reflected in the official repository of Board Policies maintained by the Office of the University Counsel pursuant to UBC's Regulatory Framework Policy.
- 11.3 <u>The Responsible Executive will normally consult with the COI Manager or the COI Committee,</u> or both, when establishing, amending, or repealing any Rules or Guidelines.
- 11.4 <u>Without limiting the generality of Section 11.1, the Rules may:</u>
 - 11.4.1 provide for unit-specific processes for reviewing disclosures, taking into consideration the following:
 - (a) the volume of disclosures generated within the academic or administrative unit; and
 - (b) <u>the need for specialized academic or professional knowledge and expertise to</u> <u>assess Conflicts of Interest or Conflicts of Commitment generated within the</u> <u>academic or administrative unit; and</u>

11.4.2 <u>establish conditional pre-approvals for defined Non-University Activities and defined</u> <u>categories of Non-University Activities that give rise to Conflicts of Interest or</u> <u>Conflicts of Commitment.</u>

T

Document comparison by Workshare Compare on Tuesday, February 13, 2024 2:06:17 PM

Input:	
Document 1 ID	file://C:\Users\ylui\OneDrive - The University Of British Columbia\Desktop\SC3 - COI - 97.docx
Description	SC3 - COI - 97
Document 2 ID	file://C:\Users\ylui\OneDrive - The University Of British Columbia\Desktop\Proposed COI Policy and Procedures being presented for Board of Governors' approval.docx
Description Proposed COI Policy and Procedures being presented Board of Governors' approval	
Rendering set	Standard

Legend:	
<u>Insertion</u>	
Deletion	
Moved from	
Moved to	
Style change	
Format change	
Moved deletion	
Inserted cell	
Deleted cell	
Moved cell	
Split/Merged cell	
Padding cell	

Statistics:	
	Count
Insertions	571
Deletions	438
Moved from	62
Moved to	62
Style changes	0
Format changes	0

Total changes	1133
---------------	------

UBC	The University of British Columbia Board of Governors	Policy No.: SC3	
Long Title:	Conflict of Interest and Conflict of Commitment		
Short Title:	COI Policy		

Background & Purposes:

The University is responsible for the advancement and dissemination of knowledge. As such, it encourages UBC Persons to seek and participate in research, to consult widely, and to engage in Non-University Activities so long as such activities do not interfere with their obligations to the University. While the University recognizes that these activities may benefit the participants, the University, and the public at large, the University is committed to ensuring that these activities are conducted in a manner consistent with the interests and mission of the University and in a manner that maintains the community's trust and confidence. Faculty Members, staff and students of the University must act with integrity and adhere to the highest ethical standards at all times.

Occasionally and as a result of normal and productive engagements inside and outside the University, Faculty Members, staff and students may find themselves in a conflict of commitment, an actual or potential conflict of interest, or in a situation where there is a perception of a conflict of interest. In fact, many beneficial and productive developments and initiatives, such as the University's commitment to liaise with industry and to transfer technology, the growth of industry-sponsored research, and the substantial increase in the University's interactions with outside organizations, also generally lead to an increase in conflicts of commitment, conflicts of interest and perceived conflicts of interest.

Conflicts of commitment, conflicts of interest and perceived conflicts of interest that go unnoticed or are improperly managed threaten to impugn the reputation and integrity of the persons involved and, potentially, the University as a whole. They undermine the public's confidence in the University's and the UBC Person's ability to pursue and disseminate knowledge, devoid of bias and personal interests. Without that public confidence, the effectiveness of the University as a public institution and of UBC Persons as intellectual leaders is diminished.

Moreover, UBC Persons are expected to vigilantly guard against conflicts of commitment, actual and potential conflicts of interest, and perceived conflicts of interest. It is important to note that the mere existence of a conflict of commitment, a conflict of interest or a perceived conflict of interest does not necessarily imply wrongdoing on anyone's part. Rather, conflicts exist regardless of a person's character, intentions and motivations. In situations where a conflict exists or there is a perception of a conflict, it is not sufficient to merely live up to one's obligations; it is necessary to do so in a timely, open, forthright, constructive and accountable manner.

Therefore, the purpose of this Policy is to: (a) identify conflicts of commitment, actual and potential conflicts of interest, and situations that may give rise to the perception of a conflict of interest; and (b) to provide procedures whereby such situations are disclosed and either avoided or appropriately managed. Management, in some circumstances, may mean avoiding the activity; and in others it may mean implementing a management protocol.

This process is not only important to preserve the integrity of the persons and processes involved, but also to ensure that the University and UBC Persons remain compliant with the requirements of the various agencies that provide research and other funding. In particular, requirements from the Tri-Council agencies, the U.S. Department of Health and Human Services, and other bodies are in place, which require the University to create and implement policies and systems to identify and manage conflicts of interest in specific ways (for example, the requirement to file an annual conflict of interest report is a requirement of the Tri-Council agencies). Failure to comply with these requirements could lead to the loss of a substantial source of research funding.

Identifying and managing conflicts of interest and perceived conflicts of interest is relevant to all research subject matters, including but not limited to research involving human subjects, and to the administration of the University.

Accordingly, all UBC Persons are responsible for informing themselves and complying with requirements imposed on them by this Policy, all related policies (the Human Research Policy), and the requirements of all agencies and organizations from which they receive funding.

Defined terms are capitalized in this Policy and can be found in Section 8 at the end of this Policy.

1. Scope

- 1.1 This Policy applies to all UBC Persons, which means full-time and part-time Faculty Members and staff members of the University, and any other person who teaches, conducts research, or works at or under the auspices of the University (including but not limited to students, adjunct and sessional faculty, librarians, program directors, post-doctoral fellows, emeriti and those holding a visiting appointment).
- 1.2 Please note, the requirement to file annual conflict of commitment and conflict of interest disclosures (as set out in Sections 3.3 and 4.4 of this Policy) only apply to those UBC Persons who are Faculty Members or who have signing authority over an active account that is used for a research purpose (see Section 1.1 of the Procedures to this Policy).

2. General

- 2.1 A Conflict of Commitment occurs where a UBC Person engages in Non-University Activities that are substantial or demanding of the UBC Person's time and attention and adversely affect the discharge of the UBC Person's responsibilities to the University. UBC Persons must disclose Conflicts of Commitment and obtain written approval prior to engaging in any activity that may give rise to a Conflict of Commitment.
- 2.2 Conflicts of Interest and perceived Conflicts of Interest can arise naturally from a UBC Person's engagement inside and outside the University, and the mere existence of a Conflict of Interest

or the perception of a Conflict of Interest does not necessarily imply wrongdoing on anyone's part. Nonetheless, Conflicts of Interest and situations that give rise to perceptions of a Conflict of Interest must be recognized, disclosed, and assessed.

- 2.3 A perceived Conflict of Interest is a situation where an actual or potential Conflict of Interest may or may not exist, but where there may be, nonetheless, from the perspective of a reasonably well-informed, impartial observer, a perception of a Conflict of Interest. This is sometimes referred to as an "apparent" conflict of interest.
- 2.4 Conflicts of Interest and perceived Conflicts of Interest fall into one of two categories: those that are permissible if appropriately managed; and those that are prohibited because they cannot be appropriately managed.
- 2.5 UBC Persons are responsible for seeking guidance from their Initial Reviewers before engaging in any activity that may be questionable. If in doubt about whether a situation would, from the perspective of a reasonably well-informed, impartial observer, appear to be a Conflict of Interest, the UBC Person must seek guidance from his or her Initial Reviewer.
- 2.6 Except as required by judicial process or law or Section 2.9 of this Policy, any information disclosed by a UBC Person to the University pursuant to this Policy will be held by the University in confidence and will only be available to those individuals who are responsible to the University for its review, for the management, administration and audit of the University, and to funding agencies where relevant to an application for research funding to that agency.
- 2.7 All UBC Persons are encouraged to bring matters of concern regarding Conflicts of Interest and perceived Conflicts of Interest to the appropriate Administrative Head of Unit. In any case, any person who knows or suspects that a Conflict of Interest or a situation giving rise to a perception of a Conflict of Interest has gone unreported, or has not been managed properly, may contact the Chair of the COI Committee, in writing, care of the Conflict of Interest Administrator, whose contact information can be found on the Office of the University Counsel's <u>website</u>. The COI Committee will review the situation in an objective, timely fashion, respecting the rights of individuals in conformity with the principles of procedural fairness in the university context, and will provide its recommendations in a report to the Responsible Executive and to the Dean of the Faculty concerned or the Administrative Head of Unit concerned, as appropriate.
- 2.8 All UBC Persons are encouraged to bring matters of concern regarding Conflicts of Commitment to the appropriate Administrative Head of Unit. In any case, any person who knows or suspects that a Conflict of Commitment has gone unreported may contact the Conflict of Interest Administrator, in writing. Upon receipt, the Conflict of Interest Administrator will forward the complaint or concern to the Initial Reviewer or such other person as the Responsible Executive may deem appropriate, who will review the situation in an objective, timely fashion, respecting the rights of individuals in conformity with the principles of procedural fairness in the university context, and will notify the Responsible Executive and the Dean of the Faculty concerned or the Administrative Head of Unit concerned, as appropriate, of the circumstances and the action(s) taken.
- 2.9 All complaints or concerns submitted pursuant to Section 2.7 and 2.8 of this Policy will be taken seriously. The anonymity of the person making the complaint or raising a concern will, unless

otherwise directed by that person, be maintained. However, any information provided anonymously will not be acted upon unless the information is independently substantiated. The University will protect personal information of all parties involved as required under the Freedom of Information and Protection of Privacy Act. The University will not tolerate any retaliation, directly or indirectly, against anyone who, in good faith, makes a complaint or raises a concern pursuant to Section 2.7 or 2.8 of this Policy, gives evidence or otherwise participates in a process under this Policy.

3. Disclosing Conflicts of Commitment

- 3.1 Where a UBC Person intends to engage in any activity that gives rise to a Conflict of Commitment, the UBC Person must disclose that activity and obtain the prior written approval of his or her Initial Reviewer.
- 3.2 The following are examples of situations where a Conflict of Commitment exists for Faculty Members, and despite the foregoing, may not be engaged in unless disclosed to and authorized by the Dean of the Faculty in which the Faculty Member has his or her primary appointment or such Dean's designate:
 - 3.2.1 Where a full-time Faculty Member's total involvement in Outside Professional Activities is greater than 52 days per year, including evenings, weekends and vacations.
 - 3.2.2 Where a full-time Faculty Member will be away from his or her place of University work for a period of 30 consecutive days (excluding holidays) as a result of the Faculty Member's Outside Professional Activities.
- 3.3 In addition, at least annually and whenever there is a material change from the previously submitted disclosure, any UBC Person identified in Section 1.1 of the Procedures to this Policy must disclose circumstances (if any) that give rise to a Conflict of Commitment, including, if applicable, the extent, nature, and timing of his or her Outside Professional Activities.

4. Disclosing Conflicts of Interest and Perceived Conflicts of Interest

- 4.1 In the University community, full and complete disclosure of Conflicts of Interest and perceived Conflicts of Interest is a key element in protecting the integrity of the University and UBC Persons and preserving the public's confidence in their ability to pursue and disseminate knowledge, devoid of bias and personal interests. Without that public confidence, the effectiveness of the University as a public institution and of UBC Persons as intellectual leaders is diminished. Therefore, UBC Persons must disclose Conflicts of Interest and perceived Conflicts of Interest to his or her Initial Reviewer and to all parties affected by the Conflict of Interest or perceived Conflict of Interest. For example, where a UBC Person makes a public statement and identifies himself or herself as being associated with the University, the UBC Person must disclose, as part of that statement, any other relevant relationships or circumstances that give rise to a Conflict of Interest or a perceived a Conflict of Interest.
- 4.2 Prior to commencing a research project, a UBC Person who is an investigator must disclose all Conflicts of Interest and perceived Conflicts of Interest that may arise in relation to the research project.

- 4.3 Prior to undertaking any activity that may give rise to a Conflict of Interest or a perceived Conflict of Interest, whether or not that activity is within the scope of the UBC Person's work at the University, a UBC Person must disclose that activity. Where prior disclosure is impossible, disclosure must be made as soon as possible after commencement of the activity.
- 4.4 In addition, at least annually and whenever there is a material change, any UBC Person identified in Section 1.1 of the Procedures to this Policy must disclose circumstances (if any) that give rise to a Conflict of Interest or a perceived Conflict of Interest.

5. Assessing and Managing Conflicts of Commitment

- 5.1 Where a UBC Person seeks approval for a Conflict of Commitment, the Initial Reviewer, Dean or Dean's delegate (as applicable) may, in his or her discretion, grant approval and may impose conditions upon such approval, taking into consideration the following:
 - 5.1.1 whether the activity interferes with the UBC Person's obligations to the University; and
 - 5.1.2 if so, whether the activity nonetheless contributes to or benefits the University in such a way that warrants the interference.
- 5.2 Where the Initial Reviewer, Dean or Dean's delegate (as applicable) determines that the activity is not permissible, the UBC Person must not proceed with the proposed activity.
- 5.3 Where the UBC Person does not agree with the decision regarding a Conflict of Commitment made by their Initial Reviewer, or in respect of a Conflict of Commitment described in Section 3.2 of this Policy, the Dean or the Dean's delegate (as applicable), the UBC Person may appeal the decision to the person holding the position immediately superior to the person who made the decision. The person who is to consider the appeal may, at his or her discretion, refuse to hear an appeal that is filed more than 30 days after the date of the Initial Reviewer's decision.
- 5.4 In addition, where the UBC Person has a joint appointment, the Initial Reviewer will be responsible to keep the other Head of Administrative Unit up-to-date with a copy of that UBC Person's disclosures and any applicable management protocol(s).

6. Role of Initial Reviewer in Assessing and Managing Conflicts of Interest and Perceived Conflicts of Interest

- 6.1 When an Initial Reviewer becomes aware of a Conflict of Interest or a perceived Conflict of Interest, he or she will assess the situation to determine what action, if any, is appropriate.
- 6.2 Where the Initial Reviewer determines that a proposed action or activity, including a proposed research project, is permissible as disclosed with no further action, the Initial Reviewer's determination will be recorded and the UBC Person may proceed with the action or activity, subject to obtaining all other applicable approvals.
- 6.3 Where the Initial Reviewer determines that an action or activity, including a proposed research project, is permissible only if appropriately managed, the Initial Reviewer will develop and

implement a protocol to manage the Conflict of Interest or perceived Conflict of Interest (as the case may be) and the UBC Person, subject to obtaining all other applicable approvals, may proceed with the proposed activity provided that he or she complies with the management protocol.

- 6.4 Where the Initial Reviewer determines that an action or activity, including a proposed research project, is not permissible, the UBC Person must not proceed with the action or activity.
- 6.5 Where the UBC Person does not agree with a decision made by the Initial Reviewer pursuant to Sections 6.1 through 6.4, the UBC Person may appeal the decision in accordance with the internal review process (if any) established under Section 5 of the Procedures or to the COI Committee, if the unit has no such internal review process. The person or committee who is to consider the appeal may, at his, her or their discretion, refuse to hear an appeal that is filed more than 30 days after the date of the Initial Reviewer's decision.
- 6.6 Where the Initial Reviewer is part of the Conflict of Interest or the perceived Conflict of Interest, or otherwise if the Initial Reviewer deems it appropriate or necessary, the Initial Reviewer will refer the disclosure to either:
 - 6.6.1 if the UBC Person concerned is a Faculty Member, his or her Dean or the Dean's delegate; or
 - 6.6.2 if the UBC person is not a Faculty Member, such other person as may be determined by the Office of the University Counsel,

and that person shall take on the role of Initial Reviewer. If the UBC Person concerned is a Faculty Member and the Faculty has an internal review process, then the Dean or the Dean's delegate shall act in accordance with such process. If it is not appropriate for either the Dean or the Dean's delegate to take on the role of Initial Reviewer, the matter will be referred to the Responsible Executive for a determination of who should take on the role of Initial Reviewer.

7. Role of the COI Committee in Assessing and Managing Conflicts of Interest and Perceived Conflicts of Interest

- 7.1 The COI Committee is charged with the following responsibilities:
 - 7.1.1 to assess disclosures of Conflicts of Interest and perceived Conflicts of Interest to determine whether, subject to obtaining all other applicable approvals:
 - (a) the proposed action or activity, including a proposed research project, is permissible as disclosed with no further action;
 - (b) the proposed action or activity, including a proposed research project, is permissible only if a protocol is implemented to ensure that the Conflict of Interest or perceived Conflict of Interest (as the case may be) is appropriately managed; or

- (c) the proposed action or activity, including a proposed research project, is not permissible;
- 7.1.2 to serve as the final avenue of appeal (such as where a UBC Person disagrees with the assessment of an Initial Reviewer or the assessment resulting from an internal process established by a Dean under Section 5 of the Procedures to this Policy);
- 7.1.3 to audit random samples of disclosures reviewed and determined to be permissible by Initial Reviewers and by those reviewing disclosures in accordance with the process established within a particular unit. The COI Committee will also periodically audit random samples of management protocols implemented in accordance with Section 7.3 below;
- 7.1.4 to educate UBC Persons on Conflicts of Interest, perceived Conflicts of Interest, Conflicts of Commitment and the application of this Policy;
- 7.1.5 to review concerns, whether self-initiated or otherwise, that a Conflict of Interest or a situation giving rise to a perception of a Conflict of Interest has gone unreported, or has not been managed properly; and
- 7.1.6 to fulfill such other roles and responsibilities assigned to it by any other policy of the University, or by the University Counsel.
- 7.2 Where the COI Committee determines that a proposed action or activity, including a proposed research project, is permissible as disclosed with no further action, the UBC Person may proceed with the proposed action or activity.
- 7.3 Where the COI Committee determines that a proposed action or activity, including a proposed research project, is permissible only if a management protocol is implemented to ensure that the Conflict of Interest or perceived Conflict of Interest (as the case may be) is appropriately managed, the UBC Person may only proceed with the proposed action or activity in accordance with the management protocol.
- 7.4 Where the COI Committee determines that an action or activity, including a proposed research project, is not permissible, the UBC Person must not proceed with the proposed action or activity.

8. Definitions

- 8.1 **"Administrative Head of Unit**" means a Director of a service unit, a Head of an academic department, a Director of a centre, institute or school, a Principal of a college, a Dean, an Associate Vice-President, the Registrar, the University Librarian, a Vice-President or the President or the equivalent.
- 8.2 **"Business**" means a corporation, partnership, sole proprietorship, firm, franchise, association, organization, holding company, joint stock company, receivership, business or real estate trust, or other legal entity organized for profit or charitable purposes, but excluding the University,

an affiliated Hospital, a private medical practice, or other entity controlled by, controlling, or under common control with the University or an affiliated Hospital.

- 8.3 **"COI Committee**" means a committee established by the University Counsel to carry out the roles set out in Section 7 of this Policy, and which includes at least one independent person who is neither a Faculty Member nor a staff member of the University.
- 8.4 "Conflict of Commitment" has the meaning given to it in Section 2.1 of this Policy.
- 8.5 **"Conflict of Interest"** means a situation in which a UBC Person, or his or her Related Party(ies), has a personal interest that conflicts or could conflict with the UBC Person's obligations to the University. Please note:
 - 8.5.1 The existence of a Conflict of Interest involves two elements: the UBC Person's or a Related Party's personal interest(s), and the UBC Person's obligations to the University. Personal interests may include business, commercial or financial interests, as well as personal matters and career interests. A UBC Person's obligation to the University is to act in the University's interests, which includes acting in support of the University's integrity and fundamental mission and avoiding circumstances that may undermine public confidence and trust. The University has many diverse objectives, which include many supporting commitments, goals and actions. From time to time these objectives, commitments, goals and actions may appear to be, or may actually be in conflict with one another. In such cases, the University's overriding interest is that its activities are undertaken with integrity and in accordance with the highest ethical standards.
 - 8.5.2 Conflicts of Interest may be actual or potential.
 - (a) An actual Conflict of Interest is a situation where the personal interest actually conflicts with the UBC Person's obligations to the University.
 - (b) A potential Conflict of Interest is a situation where the personal interest has not yet conflicted with the UBC Person's obligations to the University, but might be expected to.
- 8.6 **"Executive Position**" means any position that includes responsibilities for a material function of the operation or management of a Business.
- 8.7 "Faculty Member" means, for the purposes of this Policy, a person employed by the University as a professor, associate professor, assistant professor, professor of teaching, clinical professor, clinical associate professor, clinical assistant professor, clinical instructor, instructor I, instructor II, senior instructor, twelve-month lecturer, or in an equivalent position designated by the applicable senate.

- 8.8 "Financial Interest" means having or having the expectation to receive:
 - 8.8.1 greater than a 5% ownership interest in a single entity; or
 - 8.8.2 anything with a monetary value exceeding \$5,000 in any one calendar year, including remuneration (e.g. salary, consulting fees, retainers, honoraria, bonuses, gifts, speaker's fees, advisory board remuneration, finders or recruitment fees), equity interests (e.g. stocks, stock options or other ownership interests), and intellectual property rights (e.g. patents, copyrights, royalties or other payments from such rights).
- 8.9 "Initial Reviewer" means the individual responsible for initially reviewing a UBC Person's disclosures of Conflicts of Interest and perceived Conflicts of Interest or approving a UBC Person's Conflicts of Commitment. Except as otherwise designated by the University Counsel, a UBC Person's Initial Reviewer is the Administrative Head of Unit in which that UBC person holds his or her primary appointment.
- 8.10 **"Non-University Activity**" means any activity outside a UBC Person's scope of work with the University and includes Outside Professional Activities.
- 8.11 "Outside Professional Activity" means any activity outside a UBC Person's scope of work with the University that involves the same specialized skill and knowledge that the UBC Person utilizes in his or her work with the University and includes the operation of a Business, consulting or advisory services, external teaching, external academic appointments and external speaking engagements.
- 8.12 "Related Party" means a UBC Person's immediate family member (e.g. spouse, partner, child, parent or sibling), or other person living in the same household, any other person with whom the UBC Person shares a Financial Interest, either directly or indirectly, or any entity in which the UBC Person has an ownership interest of more than 5%.
- 8.13 "UBC Person" means all individuals identified in Section 1.1 of this Policy.

PROCEDURES ASSOCIATED WITH THE COI POLICY

Pursuant to the Regulatory Framework Policy, the President may approve Procedures or the amendment or repeal of Procedures. Such approvals must be reported at the next meeting of the UBC Board of Governors or as soon thereafter as practicable.

Capitalized terms used in these Procedures that are not otherwise defined herein shall have the meanings given to such terms in the accompanying Policy, being the COI Policy.

1. Annual Disclosures

- 1.1 At least annually, any UBC Person who is a Faculty Member or who has signing authority over an active account that is used for a research purpose, will submit a completed disclosure to his or her Initial Reviewer that discloses the following:
 - 1.1.1 all Outside Professional Activities;
 - 1.1.2 all circumstances (if any) that give rise to a Conflict of Commitment;
 - 1.1.3 all circumstances (if any) that give rise to a Conflict of Interest or a perceived Conflict of Interest; and
 - 1.1.4 such additional information asked for in the prescribed annual disclosure form, as it may be amended from time to time.
- 1.2 Annual disclosures are to be made utilizing the Conflict of Interest / Conflict of Commitment module of the Researcher Information Services (RISe) system.

2. Recognizing Conflicts of Commitment

- 2.1 The following are examples of situations where a Conflict of Commitment may exist.
 - 2.1.1 Where a UBC Person engages in Non-University Activities during his or her normal work hours at the University.
 - 2.1.2 Where a UBC Person employed on a full-time basis by the University enrols in a degree or diploma program at the University, or any other educational institution.
 - 2.1.3 Where a UBC Person uses University resources for Non-University Activities.

3. Recognizing Conflicts of Interest and Perceived Conflicts of Interest

- 3.1 As each situation depends upon its specific facts, the University has not attempted to develop an exhaustive list of Conflicts of Interest. Rather, UBC Persons must conduct themselves at all times with the highest ethical standards in a manner that will bear the closest scrutiny. The following are examples of situations where a Conflict of Interest exists:
 - 3.1.1 Where a UBC Person's responsibility to instruct and evaluate students in a fair, unbiased and effective manner is or could be impeded or compromised. The inherent power imbalance that exists between a UBC Person and a student must not be used for personal benefit.
 - 3.1.2 Where a UBC Person or a Related Party of the UBC Person has a Financial Interest in the outcome of his or her research. In the area of research, vigilance is required. The University's commitment to liaise with industry and to transfer technology, the growth of industry sponsored research, and the substantial increase in the University's interactions with outside organizations generally all lead to a corresponding increase in the potential for Conflicts of Interest. In addition, various governmental and other agencies that fund research are imposing increasingly stringent Conflicts of Interest requirements on investigators and institutions applying for research funding. UBC Persons must comply with this Policy and any applicable Conflicts of Interest policies imposed by other organizations, agencies or institutions.
 - 3.1.3 Where a UBC Person or a Related Party has a Financial Interest in their teaching activities at the University, other than their annual salary from the University.
 - 3.1.4 Where a UBC Person has influence over a decision about a proposed relationship between the University and a Business in which the UBC Person or his or her Related Party has a Financial Interest or holds an Executive Position.
 - 3.1.5 Where a UBC Person or his or her Related Party obtains a Financial Interest or an Executive Position in a Business with which the University has an existing relationship and the Business is related to the UBC Person's work at the University.
 - 3.1.6 Where a UBC Person is in a position to influence human resource decisions (such as recruitment, offer of employment, evaluation of performance, promotion, granting of tenure, or termination of employment) or admission decisions with respect to a person with whom the UBC Person has a relationship that might reasonably be perceived as creating a Conflict of Interest.
 - 3.1.7 Where a UBC Person uses his or her position with the University to solicit students, other UBC Persons, government agencies, private companies, or members of the public for Non-University Activities, including employing or soliciting employment from students and other UBC Persons for services in a personal or commercial matter.
 - 3.1.8 Where a UBC Person uses information that is acquired as a result of his or her relationship with the University and not in the public domain for Non-University Activities unless the UBC Person has proprietary rights (usually enforceable through

copyright) to that information. UBC Persons should also be aware that insider trading restrictions may also apply to them.

- 3.1.9 Where a UBC Person's obligations to a board of directors, advisory boards, or the like of an outside organization interfere with or compromise the UBC Person's obligations to the University.
- 3.1.10 Where a UBC Person accepts gifts with a value of \$500 or more that would not have been offered if not for the UBC Person's position at the University, or work within the scope of a UBC Person's employment with the University.
- 3.1.11 Where a Faculty Member enrols as a student in his or her own Faculty.
- 3.1.12 Where personal considerations compromise a Faculty Member's professional judgment in conducting or reporting research, teaching, or carrying out administrative activities.

4. Assessing and Managing Conflicts of Interest, Perceived Conflicts of Interest and Conflicts of Commitment

- 4.1 Annual disclosures will be subject to a preliminary evaluation conducted automatically against the pre-set parameters in the RISe system, which may be revised from time to time.
- 4.2 Without limiting the discretion of a person assessing a disclosure to consider all relevant factors, the following factors must be considered in assessing all disclosures:
 - 4.2.1 the impact on the UBC Person's ability to satisfy his or her obligations to the University;
 - 4.2.2 the degree to which the proposed action will be detrimental to the interests of the University, or in the research context, the degree to which it may compromise an investigator's professional judgment in conducting or reporting research; and
 - 4.2.3 the extent to which the proposed action or activity may be managed through an appropriate management protocol.
- 4.3 Where a person who reviews a disclosure is unable to determine that a proposed action or activity is permissible with no further action, the disclosure must be referred to the next level of review.
- 4.4 Where a disclosure in relation to a Conflict of Interest or a perceived Conflict of Interest is referred to the COI Committee, the COI Committee will complete a committee report for each disclosure and provide the UBC Person with a copy of the committee report.
- 4.5 Where the COI Committee determines that a management protocol is appropriate in relation to a Conflict of Interest or a perceived Conflict of Interest, the Initial Reviewer will monitor compliance with the management protocol.
- 4.6 Where the COI Committee determines that a Conflict of Interest exists, the COI Committee will:

- 4.6.1 provide the Responsible Executive with any additional information shared by the UBC Person that is not part of the disclosure;
- 4.6.2 disclose the existence of the Conflict of Interest to a funding agency, where relevant to an application for research funding to that agency; and
- 4.6.3 disclose the existence of the Conflict of Interest to the applicable research ethics board where the Conflict of Interest relates to a research project involving human participants.
- 4.7 Where the COI Committee determines that the proposed action or activity is permissible so long as the Conflict of Interest or a perceived Conflict of Interest is appropriately managed, the COI Committee may itself create, amend and/or oversee the implementation of the management protocol, or may instruct the Initial Reviewer to do so.
- 4.8 The forms of all disclosures and reports described in these Procedures will be prescribed by the Office of the University Counsel in consultation with the academic community.

5. Internal Review Process for Conflicts of Interest and Perceived Conflicts of Interest

- 5.1 A Dean, in consultation with the Office of the University Counsel, may establish an internal review process within his or her Faculty for the assessment of Conflicts of Interest and perceived Conflicts of Interest as appropriate for his or her Faculty. Generally, a level of review within each Faculty is desirable to facilitate the expeditious assessment of matters and ensure awareness of and vigilance to Conflict of Interest and perceived Conflict of Interest issues by all UBC Persons within that Faculty. For example, the appropriate process may commence with a disclosure being reviewed by the Administrative Head of Unit (as Initial Reviewer), then being referred to the Dean, and finally being referred to the COI Committee, as necessary. A Dean, in consultation with the Office of the University Counsel, may establish additional reporting requirements for Faculty Members in his or her Faculty. Provision may be made for those cases where a UBC Person disagrees with the assessment of the Initial Reviewer.
- 5.2 In establishing an appropriate internal review process, the Dean should consider the following factors:
 - 5.2.1 the volume of disclosures the Faculty generates; and
 - 5.2.2 the need for specialized knowledge and expertise in the academic/professional discipline to assess Conflicts of Interest or perceived Conflicts of Interest.

6. Guidelines for Conflicts of Commitment

- 6.1 Where an Initial Reviewer is the Head of an academic department, the Initial Reviewer may issue guidelines that address Conflicts of Commitment in his or her academic department (other than those described in Section 3.2 of the Policy), provided that such guidelines are in accordance with this Policy and Procedures. The guidelines may:
 - 6.1.1 provide for streamlined disclosure and approval procedures; and

- 6.1.2 establish conditional pre-approvals for defined activities and categories of activities that may give rise to Conflicts of Commitment.
- 6.2 Guidelines may only be issued with the prior consent of the Dean of the Initial Reviewer's Faculty.
- 6.3 Guidelines are only in effect for so long as the Initial Reviewer who issued the guidelines acts as Initial Reviewer—for greater certainty, guidelines adopted by an Initial Reviewer are no longer valid when they leave that position, and previous guidelines do not bind the succeeding Initial Reviewer.
- 6.4 The Initial Reviewer must keep an up-to-date version of the guidelines filed with the Conflict of Interest Administrator at all times.

7. Advisory Role of Office of the University Counsel

7.1 An Initial Reviewer, Administrative Head of Unit or the COI Committee may at any time seek the advice of the Office of the University Counsel with respect to any matter pertaining to this Policy, including without limitation, the establishment of a process for the assessment of Conflicts of Interest and perceived Conflicts of Interest, the assessment of any disclosure and the development of appropriate protocols for managing Conflicts of Interest and perceived Conflicts of Interest.

8. Sanctions

- 8.1 This Policy is meant to protect both the University and UBC Persons. The Administrative Head of Unit will notify the University Counsel and take immediate and appropriate action when he or she becomes aware of violations of this Policy or the Procedures.
- 8.2 Sanctions will be commensurate with the seriousness of the violation. Any disciplinary sanction that is imposed on a UBC Person pursuant to this Policy will be in accordance with the applicable terms and conditions of employment, including any applicable collective agreement, or such other contractual process(es) applicable to the UBC Person.
- 8.3 In addition to any disciplinary sanctions, where a researcher fails to comply with this Policy or these Procedures, the Initial Reviewer or the relevant Dean may direct Financial Services to freeze that researcher's research account(s), until further notice and the University may terminate funding or require the repayment of any financial benefit gained by the researcher.

Unless otherwise specified below, all section references are to sections of the proposed amended COI Policy (the "**Proposal**") set out in Appendix 1 in the <u>November 20, 2023 Submission</u> to the Board of Governors' Employee Relations Committee and all capitalized terms have the meanings set out in the Proposal.

No.	Comment	Applicable Section(s)	Committee Decision
1.	Compliance with External Agencies A commenter stated if Section 1.2 is intended to refer to the COI policies of the different agencies, this is already covered by Section 1.1.2. If it is intended to be broader, referring to any funding terms imposed by such agencies, it was suggested that this section be removed as compliance with funding terms is a requirement for all UBC Persons involved in research, not just faculty members and this is already covered in Sections 1.2 and 1.3 of the Research Policy (LR2).	1.2 of the Policy	No Changes Recommended There was consensus within the Committee to leave the language within Section 1.2 as it is currently written. It was felt that specificity in this section of the Policy was important as it serves to remind UBC Faculty Members of their responsibility to follow the requirements of ALL relevant agencies/groups/organizations when taking part in UBC related activities. It was felt that reminding Faculty members of this point was appropriate.
2.	Conflict of Commitment Example A concern was expressed around the presumption that enrolling in a degree or diploma program would be considered a Conflict of Commitment. The writer argued that while they agree that an FTE staff person could be in Conflict of Commitment if they were enrolled in a degree or diploma program and the nature of that program was such that classes, assignment, labs, seminars, exams, practicums, etc. occurred during their UBC work hours; they don't agree with the presumption that a Conflict of Commitment universally exists in this situation. <i>"There are several programs where a FTE UBC staff person could reasonably engage in a degree or diploma program without any requirement of that program interfering with the staff person's work hours/responsibilities."</i>	2.1.2 (b) of the Policy	No Changes Recommended There was consensus within the Committee to leave the language within Section 2.1.2(b) as it is currently written. It was determined that this language would ensure that engagement in these kinds of activities would require proper disclosure and approval as set forth in the Policy and would not always result in these activities being prohibited.

No.	Comment	Applicable	Committee Decision
		Section(s)	
3.	Conflict of Commitment Example A commenter suggested that Section 2.1.2(c) is an example of a COI not a COC. Non-Univeristy Activity A commenter asked, as the definition of Non- Univeristy Activity already refers to any activity outside a person's scope of work with UBC, what other activities are referred to in the fist sentence in Section 2.2.1?	2.1.2(c) of the Policy 2.2.1 of the Policy	No Changes Recommended There was consensus within the Committee that Section 2.1.2(c) more closely exemplifies a COC as using one's UBC computer, email address and work time (i.e. university resources) for Non-University Activities could adversely impact the discharge of one's primary responsibilities to UBC, a key component of the definition of a COC. There were other examples within the Policy where such behavior could also be considered a Conflict of Interest. Changes Recommended The Committee discussed this comment and decided to clarify the wording in Section 2.2.1, amending it to read as follows: S.2.2.1 "If a UBC Person intends to engage in any Non-University Activities that could lead to a Conflict of Commitment, then the UBC Person must immediately disclose that activity to their Initial Reviewer pursuant to Section 3 of the Procedures. These disclosure obligations also apply to all UBC Persons as soon as
5.	Timing of Disclosure A commenter stated that the use of the term 'arises' in the last sentence in 2.2.1 doesn't seem to align with the fact that the obligation to immediately disclose already applies when someone 'intends' to engage in an activity.	2.2.1 of the Policy	knowledge of the Conflict of Commitment arises and continue until the Conflict of Commitment ceases to exist". Changes Recommended The Committee discussed this comment and agreed that conflicts need to be disclosed as soon as knowledge of the conflict arises, not just after one has engaged in the conflict, as a COI could arise after the activity has commenced. Therefore, the text with Section 2.2.1 was further amended to emphasize the point that Conflicts of Commitment must also be disclosed as soon knowledge of the conflict arises. Section 2.2.1 was amended to read as follows: 2.2.1 If a UBC Person intends to engage in any Non-University Activities that could lead to a Conflict of Commitment, then the UBC Person must immediately disclose that activity to their Initial Reviewer pursuant to Section 3 of the Procedures. These disclosure obligations also apply to all UBC Persons as soon as knowledge of the Conflict of Commitment arises and continue until the Conflict of Commitment ceases to exist

No.	Comment	Applicable	Committee Decision
		Section(s)	
6.	Disclosing Conflicts of Commitment A commenter asked about Section 2.2.3, Wouldn't the disclosure requirement exist even if the thresholds are not exceeded? Should 2.2.3 simply read as 'they must disclose and obtain'.	2.2.3 of the Policy	Changes Recommended The Committee discussed this comment and agreed that the requirement to disclose a COC exists if a Faculty Member engages in an OPA and further more, the Faculty Member must then obtain approval of the Dean of the faculty not just the Initial Reviewer if they engage in an OPA that is greater than 52 days per year or for a period of 30 consecutive days. Section 2.2.3 was amended to read as follows: 2.2.3 If a Faculty Member intends to engage in any activity set out in Section 2.1.2(d) or Section 2.1.2(e) of this Policy, then they must disclose and obtain prior written authorization from the Dean of the faculty in which they hold their primary appointment, a copy of which must be filed with the COI Manager.
7.	Disclosing Conflicts of Interest A commenter suggested that the current wording for Section 3.2.3 seems to be already captured by Section 3.2.1. They then asked if perhaps Section 3.2.3 is intended to clarify that the PI must disclose all potential COIs (not just of the PI, but of any person involved in the research) to the UILO and ORS?	3.2.3 of the Policy	No Changes Recommended There was consensus within the Committee to leave the language within Section 3.2.3 as it is currently written. It was felt that this section of the Policy was important as it serves to highlight that the COI reporting requirements also apply to principal investigators conducting research through UBC. It was felt that emphasis made here aligned well with other Policies and was valuable to expressly state.
8.	Timing of Disclosure A commenter stated that the use of the term 'arises' in 3.2.4 doesn't seem to align with the fact that the obligation to immediately disclose already applies when someone 'intends' to engage in an activity.	3.2.4 of the Policy	Changes Recommended The Committee discussed this comment and agreed that conflicts need to be disclosed as soon as knowledge of the conflict arises, not just after one has engaged in the conflict. The text with Section 3.2.4 was amended to emphasize the point that Conflicts of Interest must be disclosed as soon knowledge of the conflict arises. Section 3.2.4 was amended to read as follows: 3.2.4 These disclosure obligations also apply to all UBC Persons as soon as knowledge of the Conflict of Interest arises and continue until the Conflict of Interest ceases to exist.

No.	Comment	Applicable Section(s)	Committee Decision
9	Personal Information A commenter suggested that in Section 4.3 of the Policy, 'any information' should instead read as 'any personal information'.	4.3 of the Policy	Changes Recommended The Committee discussed this comment and agreed with the suggested change. Making this change will mean that anonymized statistics or summaries can be compiled or released to the public or those internal to UBC who do not have a "need to know". Section 4.3 was amended to read as follows: 4.3 UBC will protect the personal information of all parties involved as required under applicable legislation, including the Freedom of Information and Protection of Privacy Act (British Columbia). Except as required by a legal process or law, any personal information disclosed by a UBC Person pursuant to this Policy will be held in confidence and will only be available to relevant granting or funding agencies and those with a need to know the information in order to fulfill their responsibilities to UBC":

No.	Comment	Applicable	Committee Decision
		Section(s)	
10	Definition of Business In relation to the definition of "Business" in Section 1.2 in the Procedures a commenter asked, "What would be an example of a medical practice controlled by UBC or an affiliated hospital / health authority that doesn't fall under the 'UBC affiliated Hospital' reference?" "Since there seem to be many private medical practices that operate out of / near the UBC affiliated hospitals, could the reference to the UBC affiliated hospitals perhaps suffice to avoid confusion? Additionally, it is my understanding that the hospitals themselves aren't legal entities. Should this perhaps read 'controlled by UBC or a UBC affiliated health authority'? "	1.2 of the Procedures	 Changes Recommended The Committee discussed this comment and decided to remove the exception of UBC affiliated hospitals or entities controlled by UBC as follows: "Business" means a corporation, partnership, sole proprietorship, firm, franchise, association, organization, holding company, joint stock company, receivership, business or real estate trust, or other entity organized for profit, not for profit, or charitable purposes. This change was made for the following reasons: Sponsored research affiliations with hospitals are getting more complex and the affiliation agreements are being renewed and renegotiated (i.e., health authorities wanting ownership over the research funds, etc.) It is not necessarily prohibited by having a person disclose their clinical affiliation or their role with another UBC entity if only to remind them to behave appropriately when wearing separate hats. (i.e., clinical faculty giving an expert opinion on UBC letterhead – inappropriate) Erring on the side of over disclosure rather than under disclosure helps decision makers make more informed decisions about COI/COCs, particularly with more complex fact scenarios This will change will require extra training for those who have not been traditionally disclosing certain arrangements as part of a Business but this can happen as part of the new COI Policy roll out, etc.
11	Definition of UBC Person(s) A commenter asked why the definition of 'UBC Person' in SC3 is different from the definitions of 'UBC Person(s)' used in other policies, such as the definition used in the Research Policy (LR2). What is the reason for the difference?	1.24 of the Procedures	No Changes Recommended There was consensus within the Committee for leaving the definition of UBC Person(s) as currently written. Section 2.1 in the Regulatory Framework Policy (GA2) allows flexibility in the definition of those individuals to whom a Policy applies. Definitions provided for UBC Persons (in SC6, LR2 & SC3) University Person (in LR11) UBC Member (in SC18) and Members of the UBC Community (in SC7 & SC17) all slightly differ in their presentation.

No.	Comment	Applicable	Committee Decision
		Section(s)	
12	Collectives A concern was expressed that the amended COI/COC Policy does not address the issue of "Collectives" that pursue research and related goals together. The commenter stated that Collectives present possible COI/COC issues. "For example: One member of the collective, who is external to UBC, applies for a faculty position and another member of the collective is a UBC faculty member on the hiring committee. They are not related parties and share no financial interest. However, they share a common interest in the "collective" they both belong to. This represents a conflict of interest very similar to that which related parties or financial interests pose. Yet the current COI/COC would not require the parties to declare or manage this conflict." The commenter suggests that it is past time for the university to develop a policy on "Collectives." In the absence of this, they believe	2.1 of the Procedures	 No Changes Recommended The Committee discussed this comment and decided that Collectives appear to be a type of collaborative relationship that could be used to pursue specific research interests, for example: Research Collectives facilitate interdisciplinary work by scholars whose interests range across disciplines and topic areas. Research Collectives seek to bring together several disciplinary perspectives to promote a more integrative understanding of the science of selected topics. Research Collectives consolidate and further develop research expertise across Departments, Faculties, Universities and elsewhere, facilitating debates, synthesis and the mobilization of knowledge between professors at all ranks, students of all levels and community members The Committee felt that such relationships are already addressed within SC3. While "Collectives" are not cited by name, Sections 2.1.7 and 2.1.13 in the Procedures would appear to apply to these kinds of relationships, as the list of COIs under Section 2.1 of the Procedures is not meant to be an exhaustive list.
13	the COI/COC policy should address "Collectives". Transferring Research Results A commenter suggested that we should avoid the implication that a UBC Person has the right to transfer research results or intellectual property arising from their UBC research to a Business in which the UBC Person or a Related Party has a Financial Interest or holds a Position of Authority. It was suggested that this may contradict the Invention Policy (LR11).	2.1.6 of the Procedures	No Changes Recommended There was consensus within the Committee for leaving the language within Section 2.1.6 as originally written as it is a COI regardless if the IP was transferred correctly or not.

No.	Comment	Applicable	Committee Decision
	connent	• •	
14	Recruitment/Hiring A commenter suggested that the policy does not make sufficient (or any) mention of COI/COC that arise in the recruitment and hiring process.	Section(s) 2.1.7 of the Procedures	Changes Recommended The Committee discussed this comment and decided that Section 2.1.7 in the Procedures does address this issue but that the examples provided within this Section should be expanded by adding "appointment or selection process" and as well as should include reference to investigations and disciplinary measures. Section 2.1.7 was amended to read as follows: 2.1.7 where a UBC Person is in a position to influence human resource decisions (such as recruitment, offer of employment, appointment or selection process, evaluation of performance, promotion, granting of tenure, investigations, discipline, or termination of employment) or admission decisions with respect to a person who is a Related Party or who has another relationship with the UBC Person that might reasonably be perceived as creating a Conflict of Interest;
15	Timing of Disclosures A commenter suggested that the language in 3.1.1 in the Procedures seems to contradict the provisions in the Policy that require immediate disclosure when someone 'intends' to engage in an activity.	3.1.1 of the Procedures	 Changes Recommended The Committee discussed this comment and agreed that there needs to be disclosure both prior to engaging in an activity and also if while engaged in an activity a COI arises. While this is already clarified in Sections 2.2.1 and Section 3.2.4 of the Policy, the Committee decided to add the following language to Section- 3.1.1 (Required Disclosure) of the Procedures to further clarify, pushing the original Sections down to 3.1.2 and 3.1.3 respectively: 3.1.1 immediately disclose to their Initial Reviewer if a UBC Person intends to engage in any Non-University Activities that could lead to a Conflict of interest and Conflict of Commitment;

No.	Comment	Applicable	Committee Decision
		Section(s)	
16	Non-University & Outside Professional Activities A commenter asked "what is the reason that Staff Members have to disclose any Non- University Activities, while Faculty Members only have to disclose Outside Professional Activities?"	3.3.1 of the Procedures	 Changes Recommended The Committee discussed this comment and decided that there should be consistency between disclosures of both the Staff Members and the Faculty Members and the definition of Non-University Activities includes Outside Professional Activities, therefore the Committee agreed to amend Sections 3.2 through 3.5 in the Procedures to now read as follows: 3.2 At least annually, all Faculty Members and Staff Members must submit a completed disclosure that discloses the following: 3.2.1 all Non- University Activities; 3.2.2 all circumstances (if any) that give rise to a Conflict of Interest or Conflict of Commitment; and 3.2.3 such additional information asked for in the prescribed annual disclosure form, as it may be amended from time to time. 3.3 All disclosures made by Faculty Members under Section 3.2 of these Procedures must be submitted utilizing the Conflict of Interest or the Conflict of Commitment module in the <i>RISe</i> system. 3.4 All disclosures made by Staff Members under Section 3.2 of these Procedures must be submitted utilizing the Conflict of Interest or the Conflict of Commitment module in the <i>Workday</i> system.
17	COI Manager A commenter asked if the reference in Section 6.1.4 in the Procedures should be to Section 4.3 instead of Section 4.1?	6.1.4 of the Procedures	No Changes Recommended The Committee discussed this comment and indicated that the reference is to Section 4.1 in the Policy, not the Procedures as the commenter suggested, and that the reference is correct.

No.	Comment	Applicable	Committee Decision
		Section(s)	
18	Research Accounts	10.2 of the	Changes Recommended
	A commenter stated that for Section 10.2 in the	Procedures	The Committee discussed this comment and recommend amending 10.2 to
	Procedures, research accounts may also be		include the UILO and to clarify that UBC does not terminate funding but rather
	administered by the UILO.		may terminate access to funding as follows:
			10.2 In addition to any disciplinary action, where a Faculty Member
			fails to comply with the Policy, the Dean or the Initial Reviewer (if the
			Initial Reviewer is not the Dean) may direct either the Office of Research
			Services or the University - Industry Liaison Office to freeze the research
			account(s) of the Faculty Member until further notice. UBC may also
			terminate access to funding or require the repayment of any financial
			benefit gained by the Faculty Member.